SEPTEMBER 2024

Power Plays

Europe's Response to the Energy Crisis

AUTHORS Max Bergmann Cy McGeady Otto Svendsen Mathias Zacarias Ignacio Urbasos

A Report of the CSIS Europe, Russia, and Eurasia Program and the CSIS Energy Security and Climate Change Program



Power Plays

Europe's Response to the Energy Crisis

AUTHORS Max Bergmann Cy McGeady Otto Svendsen Mathias Zacarias Ignacio Urbasos

A Report of the CSIS Europe, Russia, and Eurasia Program and the CSIS Energy Security and Climate Change Program



About CSIS

The Center for Strategic and International Studies (CSIS) is a bipartisan, nonprofit policy research organization dedicated to advancing practical ideas to address the world's greatest challenges.

Thomas J. Pritzker was named chairman of the CSIS Board of Trustees in 2015, succeeding former U.S. senator Sam Nunn (D-GA). Founded in 1962, CSIS is led by John J. Hamre, who has served as president and chief executive officer since 2000.

CSIS's purpose is to define the future of national security. We are guided by a distinct set of values—nonpartisanship, independent thought, innovative thinking, cross-disciplinary scholarship, integrity and professionalism, and talent development. CSIS's values work in concert toward the goal of making real-world impact.

CSIS scholars bring their policy expertise, judgment, and robust networks to their research, analysis, and recommendations. We organize conferences, publish, lecture, and make media appearances that aim to increase the knowledge, awareness, and salience of policy issues with relevant stakeholders and the interested public.

CSIS has impact when our research helps to inform the decisionmaking of key policymakers and the thinking of key influencers. We work toward a vision of a safer and more prosperous world.

CSIS does not take specific policy positions; accordingly, all views expressed herein should be understood to be solely those of the author(s).

© 2024 by the Center for Strategic and International Studies. All rights reserved.

Center for Strategic & International Studies 1616 Rhode Island Avenue, NW Washington, DC 20036 202-887-0200 | www.csis.org

Acknowledgements

The authors are thankful to several colleagues who provided feedback on early drafts of this report. They include Joseph Majkut and Federico Steinberg. This report could not have been written without their significant contributions. Finally, we owe a debt of gratitude to the CSIS iDeas Lab and communications team, especially Katherine Stark, Phillip Meylan, and Lauren Bailey, for their help in the editing and publication of the report.

This report is made possible by the generous support of the Public Power Corporation S.A.

Contents

Executive Summary	V
Introduction	1
1 Before the Storm: Historical Perspective on European Energy Strategy	5
2 The Crisis: Fall 2021 through Winter 2022-23	16
3 The Response: Policy Interventions and Policy Costs	28
4 Postcrisis Policy Landscape: Toward a New European Energy Security Strategy?	48
5 Lessons Learned	60
Conclusion	63
About the Authors	64
Endnotes	66

Executive Summary

R ussia's full-scale invasion of Ukraine in February 2022 triggered a severe energy crisis in Europe. The shock represented a test not just of Europe's energy sector but of the resilience of Europe's democratic political system and market economies. Europe's strong response has demonstrated that an accelerated energy transition is possible amid strong political and economic headwinds when backed by a broad popular consensus for ambitious action.

The results of Europe's actions have been remarkable. The European Union has reckoned with the false promise of its long-standing energy interdependence with Russia, as the energy sector has been central to Putin's consolidation of power inside of Russia and has been weaponized in the Kremlin's engagement with neighboring countries. In 2022, Russian fossil fuels supplied 1 in 5 units of energy consumed in the European Union; in 2024, that figure fell to 1 in 20.¹

As imports of Russian fossil fuels fell precipitously after Russia's invasion, EU member members took ambitious steps to enhance domestic energy supply capacities and develop alternatives to Russian imports. Coal-fired power plants were temporarily revived, policies to phase out nuclear power were revisited, Norwegian piped gas flows into the European Union increased significantly, transatlantic coordination redirected liquified natural gas volumes toward Europe, and renewables were dramatically expanded. EU institutions were instrumental in overseeing a unified response and set the direction for its member states, from creating collective agreement on the need to decouple from Russian energy over time to setting targets for demand reduction and ensuring that Europe would not abandon its climate goals.

While Europe emerged from the crisis as a stronger energy actor, challenges remain in areas spanning electricity market reform to hydrogen diplomacy to protecting critical undersea infrastructure. From a supply perspective, Europe has found a way to replace Russian gas. But this energy is largely coming from sources external to the European Union, and the costs of these alternative sources will be higher and therefore impact Europe's economic competitiveness. Furthermore, as Europe accelerates its green transition, it must address the risk posed by increasing import dependencies in clean tech supply chains, particularly from China.

The crisis spawned many lessons relevant for experts and policymakers in a variety of fields. Most of the lessons are positive. Europe's democracies and markets proved resilient to shocks, Russia failed in its attempt to blackmail EU member states into ceding support for Ukraine, and the European Union proved its geopolitical mettle amid a potentially devastating crisis. However, the crisis also left a significant fiscal mark on European economies and, through price shock and inflationary effects, gave rise to more anti-establishment parties, many of which may seek to roll back aspects of the energy transition.

Introduction

R ussia's full-scale invasion of Ukraine in February 2022 triggered an energy crisis in Europe unmatched in severity since the oil shock of the 1970s. The energy shock became a test not just of Europe's energy sector but of Europe's union and the resilience of its democratic political system, market economy, and free and open societies. How Europe responded to this crisis explains much about the European project, the energy transition, and the nature of energy security in a new era of profound geopolitical competition and upheaval. Europe has demonstrated that a rapid energy transition is possible in the face of an urgent need and when backed by strong political consensus.

In the more than two years since the start of the war, Europe's energy mix and energy map have changed dramatically. Europe today appears not only more energy resilient than before the crisis, but also just as intent on achieving an energy transition that will be responsive to climate change and that will reduce its external dependencies and improve competitiveness.

The results of Europe's actions have been remarkable. The European Union has reduced Russian fossil fuel imports from \$16 billion per month in early 2022 to around \$1 billion per month by the end of 2023, with cuts to oil imports representing the main driver of the decline.² In 2022, Russian fossil fuels supplied 1 in 5 units of energy consumed in the European Union; in 2024, that figure fell to 1 in 20.³



U.S. assistant secretary for energy resources Geoffrey R. Pyatt providing opening remarks at the 2024 CSIS Energy Security and Geopolitics Conference.

"Russia has decisively proven itself to be an unreliable supplier of energy, that will choose to weaponize its energy resources as a tool of political coercion... Today we're in an environment where Russia is never again going to be seen as a reliable supplier of energy. And where Europe's transition away from Russian energy has progressed much faster than predicted and marks a really permanent shift in the international energy map."

- Geoffrey R. Pyatt, U.S. Assistant Secretary for Energy Resources⁴

This is not to say that European leaders did everything right or that this transition was inexpensive. The loss of Russian supplies amid the war disproved decades of belief that interdependence would secure peace. Disbelief in Europe's capacity to manage the crisis may also have prevented even bolder action against the Russian energy sector in the early stages of the war, blunting the effects on the Russian economy. Even today, Russian liquified natural gas (LNG) and piped gas are still part of Europe's energy mix, and Europe remains dependent on Russia for nuclear fuel.

Massive fiscal outlays were deployed to keep industry afloat, particularly in Germany. According to Bruegel estimates, from the beginning of the crisis in September 2021 until June 2023, €651 billion was allocated and earmarked across European countries to shield consumers from rising energy costs.⁵ The loss of a relatively cheap and seemingly stable supply of Russian natural gas has significant implications for industrial competitiveness, and for the German economy in particular. Longer-term issues related to persistently elevated prices and the implications for the European Union's competitiveness will still have to be addressed.



Georgios Stassis, chairman and CEO of Public Power Corporation S.A., providing opening remarks at the 2024 CSIS Energy Security and Geopolitics Conference.

"There is no doubt that geopolitics and political stability is a prerequisite for energy security, and vice versa. This was a hard lesson we learned in Europe from the recent geopolitical and energy crises and came at high cost. It pushed us to reconsider the definition of security of supply."

- Georgios Stassis, Chairman and CEO of Public Power Corporation S.A., Greece⁶

In 2022, there were fears that the shock of energy costs would turn Europeans against supporting Ukraine; however, these were unfounded. As European households turned down their thermostats in the winter of 2022-23 and paid, on average, 14 percent more for electricity and 114 percent more for gas to heat their homes from the first half of 2021 to October 2022, they knew Russia was to blame for the crisis, and support for Ukraine remained high.⁷

The European project has advanced in this crisis. European governments worked together through the European Union to adopt sanctions and construct a crisis energy plan. The European Union leveraged the resources it borrowed in response to the Covid-19 pandemic to help countries cope and established an effort to procure LNG on behalf of the bloc. As a result, the European Union is a much more prominent energy actor than before the crisis. The situation demonstrates not only that the European Union is forged in crisis but also that the union enables Europe to respond to crises much more powerfully and effectively than if countries operated independently.

Transatlantic relations also strengthened as U.S.-EU cooperation on energy and sanctions significantly increased. U.S.-EU engagement and coordination in these sectors have become much

more robust and have laid the groundwork for a stronger U.S.-EU relationship—a relationship that historically has been strained or limited.

This report reviews Europe's response to the energy shock following Russia's invasion of Ukraine. Chapter 1 sets the stage by examining the European energy system and approach to energy security prior to the invasion. Chapter 2 outlines how Russia weaponized energy and details the challenges posed to Europe's energy supply in 2022. Chapter 3 outlines Europe's response to the crisis, including the steps it and its partners have taken, and its effort to decouple from Russian energy. Chapter 4 examines the postcrisis policy landscape and identifies current and future challenges for Europe. Chapter 5 offers some thoughts on lessons learned from the crisis. Chapter 6 provides some concluding thoughts.

Before the Storm

Historical Perspective on European Energy Strategy

he link between energy and security has been at the heart of the European project since its inception after World War II. The creation of the European Coal and Steel Community (ECSC) in 1952–involving France, West Germany, Italy, Belgium, the Netherlands, and Luxembourg–marked the first attempt at integrating European markets after the war. Coal and steel were the focus because these sectors were critical to waging war. Robert Schuman, the French foreign minister and the ECSC's principal architect, noted that "the solidarity in production" from pooling coal and steel production would make "any war between France and Germany . . . not merely unthinkable, but materially impossible."⁸ As a catalyst of European integration, energy is core to the conception of the European Union as a project that brought peace to Europe.

Yet, while energy was an initial focus for European integration, it was largely pushed to the side as European integration marched forward through the decades. Although energy market integration was continuously discussed over the following decades, the focus for policymakers was squarely on the security of supply.⁹ That meant European energy integration made limited advances relative to the extensive economic and political integration Europe would experience; energy policy would remain the domain of Europe's national governments.

It was not until the 1980s, when policymakers shifted their focus to forging a single European market, that European energy integration was considered. A 1988 European Commission working paper conceptualized the idea of a functioning internal energy market, recognizing the need to harmonize rules and technical norms, liberalize public procurement, and remove fiscal barriers such as diverging tax laws.¹⁰ However, the recommendations set forth in the working paper were

met with widespread opposition, as they sought to uproot markets with entrenched incumbents enjoying both national and natural monopolies. Although the commission continued developing plans for greater energy integration and liberalization in the 1990s, once again no energy chapter was included in the 1992 Maastricht Treaty, which formed the European Union. Member states vetoed efforts to include energy in the treaty, wishing to retain control over their national energy policies.

In the following decades, the European Union took incremental steps to create an internal energy market. Between 1996 and 1998, the European Parliament passed its First Energy Package directives on the electricity and gas markets, which were weaker versions of the commission proposals that member states in the European Council had rejected.¹¹ However, according to Joseph Dutton, "Nevertheless they substantially bolstered the moves towards an internal marketplace and attempted to remove 'legal monopolies' and obliged vertically integrated companies to grant third party access to networks."¹² Yet the range of transmission systems and lack of crossborder infrastructure linkages hindered formation of an internal European market. In the 2000s, environmental policy became much more of a focus, and the European Union continued to pass regulations to address friction in the internal market. In 2019, the European Union adopted its Fourth Energy Package, which strengthened the role of the EU Agency for the Cooperation of Energy Regulators in facilitating cross-border cooperation.¹³

Despite some progress, Europe entered the 2022 energy crisis, 50 years after the Middle East oil embargoes of the 1970s, far from achieving a true energy union. From a regulatory point of view, the energy sector had been liberalized by eliminating monopolies, implementing unbundling rules, and opening access to infrastructure to competitors. However, implementation of these rules often had limited success due to strong resistance. Legacy monopoly companies still had a dominant market position, and the development of interconnections, both gas and electricity, was in many cases insufficient to guarantee an integrated and functional cross-border market.¹⁴

The False Promise of Energy Interdependence with Russia

Following the collapse of the Soviet Union, the main threat to European security was seemingly gone. Not only could Europeans enjoy a peace dividend and shrink their defense budgets, but engagement with Russia could seemingly strengthen Europe's energy security as well. There was vast potential in the Russian energy sector, not only to strengthen links with Europe's advanced economies but also to invest and modernize these Russian industries.

Drawing inspiration from Europe's postwar experience, the hypothesis of change through trade (*Wandel durch Handel*)–mostly associated with German foreign policy–gave a strategic rationale for deepening European energy integration with Russia.¹⁵ EU leaders believed interdependence would bring trading partners closer to European values and practices. Thus, buying Russian energy and investing in the Russian energy sector not only were good for Europe economically but also would serve to keep Russia on the democratic path.

The flaws in the "change through trade" strategy became apparent well before Russia's 2022 full-scale invasion of Ukraine. The energy sector has been central to Putin's Russia since he came to office in 2000; it was essential to his consolidation of power inside Russia and a key part of Russian foreign policy. Upon becoming president in 2000, Putin sought to exert control over the energy sector, leading to confrontations with Russian oligarchs such as the head of Yukos, Mikhail Khodorkovsky, who was arrested after questioning the illiberalism of the Putin regime. Putin has personally selected the heads of Russian energy companies, such as Igor Sechin, CEO of Rosneft. There are numerous accounts of Putin demonstrating to Western officials and businesspeople his in-depth understanding of energy markets. For example, Daniel Yergin writes in *The New Map* that Putin "understood the power that came from Russia's oil and gas. Western interlocutors would be consistently surprised by his detailed knowledge of the energy industry and energy markets and the fluency with which he discussed the intricacies—as much like a CEO, they would say, as a head of state."¹⁶

Russia repeatedly weaponized energy resources in its engagement with neighboring countries. Gazprom threatened to cut the supply of gas to Ukraine in 2005 when Ukraine refused to accept a fivefold price increase.¹⁷ Prior to Russia's invasion of Georgia, disruptions in gas flows forced the Georgian government to ration supplies.¹⁸ In the week following Russia's 2014 invasion of Crimea, Gazprom hiked the price of gas exports to Ukraine from \$268.50 per thousand cubic meters (mcm) to \$385.50 per mcm, and ultimately to \$485.00 per mcm.¹⁹ Russia has also provided gas cost-free to the breakaway region of Transnistria in Moldova since 2005.

Russia's energy blackmail of former Soviet countries failed to induce a serious reassessment of European energy security. On the one hand, Russia's tactics were not seen as applicable to Europe given its market size. On the other hand, many European leaders and businesses struggled to see an alternative, especially since U.S. LNG exports to Europe were insignificant in volume, renewables were still largely aspirational, nuclear power suffered from deep negative connotations, and coal had stark climate ramifications.

Europe's faith in mutual interdependence, however, failed to recognize that Putin's Russia had consistently prioritized geopolitical advancement over the health of its economy. Yet Russia believed Europe's priorities were the opposite: European leaders and societies would put their economic well-being ahead of geopolitics. The long-held Soviet-era perception of the decadent, capitalist West caring only about profit remains deeply rooted in the Kremlin today. As such, as Russian forces amassed on Ukraine's borders, Putin likely believed he had leverage. Russia could absorb the economic and political hit caused by an energy shock, whereas European leaders could not; therefore, Europe's response to the invasion would be performative and limited.

Development of European Gas Markets and Integration with Russia

The European Union has consistently been dependent on energy imports, particularly oil, petroleum products, and natural gas.²⁰ The data shows that energy import dependency has grown since 1990 (see Figure 1). In 2021, 91.7 percent of EU energy demand for oil and petroleum products

and 83.4 percent of demand for natural gas were covered by extra-EU imports. By 2021, a significant share of EU imports came from Russia, including 45 percent of total gas imports and nearly half of total coal imports.²¹ At that time, more than half of Russia's oil exports were also directed to Europe, which accounted for about one-third of Europe's total oil imports.²²



Figure 1: EU Energy Dependency by Fuel, in Selected Years 1990-2022 (Petajoules)

Source: "Energy statistics - an overview," Eurostat, May 2024, https://ec.europa.eu/eurostat/statistics-explained/index.php? title=Energy_statistics_-_an_overview#Energy_dependency.

In this context, the gas infrastructure connecting Russia to Europe has been the clearest symbol of Russia's status as a fossil fuel superpower and its influence over Europe. The geopolitics of these gas pipelines has historically been complex for the European Union and its member states, given their diverging interests and outlooks as well as historical ties to Russia (the first pipelines date to the 1960s). After the Cold War, and thanks to an effective commercial diplomacy campaign, Gazprom established new infrastructure projects (TurkStream and Nord Stream), which made it possible to circumvent transit countries identified as problematic–mostly Ukraine and Poland.

No energy infrastructure project reflected this development better than the Nord Stream gas pipelines connecting western Russia with Germany via the Baltic Sea. Established by a consortium led by Gazprom and operational since 2011, Nord Stream 1 had a capacity of 55 billion cubic meters (bcm), equivalent to roughly a third of EU gas imports from Russia in 2021.²³

Russia's first invasion of Ukraine occurred in March 2014 when it annexed the Crimean Peninsula. Despite these evolving security and geopolitical realities, Gazprom, Royal Dutch Shell, E.ON, OMV, and Engie reached a commercial agreement roughly a year later over Nord Stream 2, a second gas pipeline connecting Russia and Germany. However, extensive legal battles and U.S. sanctions beginning in 2019 significantly delayed construction of the pipeline.

Backers of the pipeline projects, which included the Russian government and high-profile German politicians from the Christian Democratic Union and the Social Democratic Party argued that they would ensure stability of supply by connecting Europe with the world's largest gas reserves in Russia while supporting the continent's sustainability goals as a less carbon-intensive complement than coal to renewable energy.²⁴

Opponents, which included the United States, various environmental groups, several other German politicians, and many of the Eastern European countries most exposed to Russian aggression, claimed that the projects would lock in dependence on Russia, enable Russia to extort Ukraine, be harmful to marine ecosystems, and undermine Europe's green transition.²⁵ Despite attempts by supporters to frame the projects as a strictly commercial undertaking, opponents argued that an overreliance on Russian gas would leave Europe at the Kremlin's mercy and susceptible to energy blackmail.



Anna Mikulska, a research staff member at the Institute for Defense Analyses, speaking at the 2024 CSIS Energy Security and Geopolitics Conference.

"You should probably underscore that Europe is not as uniform as we might often think.... There's the UK versus EU, but there's also the Western versus Eastern Europe, or the post-Soviet Europe, which in my view has seen energy security, from the perspective of dependence on Russia, very differently than the West."

— Anna Mikulska, Research Staff Member, Institute for Defense Analyses²⁶

The Nord Stream pipelines were thus largely a German story. Germany is the largest EU country in terms of the size of its population and economy. Its economic weight is disproportionate, representing a quarter of overall EU gross domestic product (GDP). While efforts were made (and sometimes are still made) to portray Nord Stream 2 as a European project, it was driven by German commercial, economic, and political interests. During a visit to St. Petersburg in November 2017, German foreign minister Sigmar Gabriel affirmed German support for the project. He stated, "Now is a good time for us to intensify our economic cooperation. We've always known Russia as an especially reliable gas supplier."²⁷

Nevertheless, the interconnected European gas network and markets meant that Russian natural gas could also make its way to France, the Netherlands, and other European states and influence their energy systems. The availability of low-priced supply made it easier for the Netherlands to close its Groningen gas field given local earthquake concerns. While the United States undertook a shale gas revolution, European gas production declined dramatically in the last decade, in large part due to the perceived stability and low cost of Russian supply.²⁸ Private investment in European gas production made little sense given the economic headwinds of Russian gas and the political headwinds of climate policy.



French prime minister Francois Fillon, German chancellor Angela Merkel, Dutch prime minister Mark Rutte, Russian president Dmitry Medvedev, and EU energy commissioner Guenther Oettinger inaugurate the Nord Stream Baltic Sea gas pipeline on November 8, 2011.

Photo: Sean Gallup/Getty Images

Meanwhile, Germany's broader energy policies compounded the energy security risks of a strategic bet on Russian piped gas. Germany's decision to phase out nuclear power stands out. While nuclear power supplied a quarter of Germany's electricity demand a decade ago, today no nuclear power plants remain operational.²⁹ Availability of cheap Russian gas enabled the economic feasibility of this nuclear phaseout as gas generation and renewables replaced lost nuclear generation capacity.

The large contribution by nuclear power was controversial, as public support for it has eroded since the 1980s. By 2002, public pressure brought forth a law mandating the closure of Germany's entire fleet of nuclear power plants. The 2011 Fukushima accident unleashed significant blowback on German chancellor Angela Merkel, until then a staunch supporter of nuclear power, and catalyzed previously stalled implementation of the 2002 law. For example, Merkel ally and long-time defender of atomic energy Stefan Mappus lost the state premier position as an incumbent in Baden-Württemberg, an otherwise conservative state, to a challenger from the Green Party.³⁰ Three months later, the Bundestag agreed to phase out nuclear energy by 2022. Although the shutdown was delayed in October 2022 for several months due to energy security concerns related to the war in Ukraine, Germany's last nuclear power plant disconnected from the electricity grid in April 2023.³¹





Source: "Simplified energy balances," Eurostat, May 2024, https://ec.europa.eu/eurostat/databrowser/view/nrg_bal_s_custom_ 11587872/default/table?lang=en.

By 2020, Russian gas pipelines supplied 41.5 percent of total EU gas imports, with a dominant position in many captive markets through long-term contracts and situations of quasi-monopoly (see Figure 2). Gazprom's influence also extended to the commercialization and storage of its gas on European soil. Through its subsidiaries (Gazprom Germania and Gazprom Marketing and Trading), it managed a dense network of underground gas storage facilities, distribution networks, and trading companies. In 2020, Gazprom Germania had total assets of €8.4 billion and equity of €2.2 billion.³² In particular, Gazprom Germania had the largest gas storage capacity in the European Union, with infrastructure in Austria, the Czech Republic, Germany, and the Netherlands, which gave the company a dominant position in the EU gas market. Gazprom's influence was not limited to energy. The company's clout extended from politics to sports through funding and sponsorship, making it a recognizable brand to many European citizens.



Figure 3: Major Pre-Invasion Russian Gas Pipelines to Europe

Source: "Global Gas Infrastructure Tracker," Global Energy Monitor, June 2024, https://globalenergymonitor.org/projects/global-gas-infrastructure-tracker/tracker/.

The United States has exerted significant diplomatic pressure on Germany and consistently framed the Nord Stream projects in geopolitical terms. In December 2019, U.S. president Donald Trump signed a law that would impose sanctions on any firm involved with helping Gazprom finish the pipeline into the European Union, describing the project as a "tool of coercion."³³ Even then, the U.S. Department of State highlighted the adverse implications for Ukraine, noting that Nord Stream 2 "would enable Russia to bypass Ukraine for gas transit to Europe, which would deprive Ukraine of substantial transit revenues and increase its vulnerability to Russian aggression."³⁴ However, critics of Washington's approach often interpreted U.S. geopolitical pressure as partly a cynical effort to advance the U.S. LNG industry.³⁵ The administration of U.S. president Joe Biden continued opposition to Russian gas. In March 2021, Secretary of State Antony Blinken described the pipeline as "a Russian geopolitical project intended to divide Europe and weaken European energy security."³⁶

Nord Stream 2 was one of the most prominent issues in Chancellor Merkel's final visit to Washington in July 2021.³⁷ On that visit, President Biden agreed not to sanction German companies participating in the project, stating, "Good friends can disagree . . . but by the time I became President, it was 90 percent completed. And imposing sanctions did not seem to make any sense."³⁸ This was seen as a major concession to the chancellor and drew significant political blowback from both sides of the aisle in Washington.³⁹ By September 2021, the construction of Nord Stream 2 had been completed, awaiting only certification from German regulators to begin operating.⁴⁰

But this was not simply a case of the dependence of Germany, Europe's largest economy. Given its position in Europe, Germany was effectively imposing dependence on the rest of Europe. Eastern European countries, whose energy systems were built to integrate with Russia's, were also heavily dependent on Russian energy supplies. Poland, one of the most vocal countries about the threat posed by Russia, received 46 percent of its natural gas, 64 percent of its oil, and 15 percent of its coal from Russia in 2022, according to the think tank Forum Energii.⁴¹ Latvia was 100 percent dependent on Russia for its natural gas supplies.⁴² Changing course in Europe's broader energy strategy would become too costly.

Thus, as more than 100,000 Russian forces surrounded Ukraine in the autumn of 2021, Europe found itself utterly dependent on Russia for its energy security. For example, at the time Germany lacked any LNG import capacity. In 2022, Germany imported 55 percent of its natural gas from Russia, and with Nord Stream 2 set to open, that dependence was likely to increase.⁴³

Precrisis Energy Transition Plans

In the decade before Russia's invasion, European energy policy prioritized the energy transition over energy security considerations. In 2019, the European Commission unveiled the European Green Deal, organized around the goal of climate neutrality by 2050.⁴⁴ The new climate strategy sought to transform the European Union into a "modern, resource-efficient and competitive economy."⁴⁵

Implementation of the Green Deal's ambitious climate objectives in the energy sector took the form of the Fit for 55 package presented by the European Commission in July 2021.⁴⁶ The package translated into law the Green Deal's goal of reducing EU-wide emissions by 55 percent by 2030. This package increased renewable energy targets, approved the Carbon Border Adjustment Mechanism, and expanded the scope of the EU Emissions Trading System.⁴⁷

Conspicuously missing were provisions advancing energy security. Arriving just prior to the start of the crisis, these policies effectively demonstrated the EU policymaking community's underweighting of energy security in favor of the energy transition.

DEBATES OVER NUCLEAR, GAS, AND COAL: PHASEOUTS AND GREEN TAXONOMY

The electricity-generation resource mix naturally impacts policy considerations in both climate and energy security. However, precrisis European policy debates on the subject primarily revolved around climate. Policymakers generally favored renewable energy expansion in the form of wind and solar as the primary locus for energy investment and as the primary vehicle for Europe's green transition. Implicit in this model was a large, albeit eventually declining, role for natural gas as a balancing resource.

During the first half of 2021, as Russia was laying the groundwork for its invasion, European energy policymakers were engaged in a fierce political debate over the EU taxonomy for sustainable activities.⁴⁸ This labeling system, commonly referred to as the "green taxonomy," guides investors toward economic activities that are compatible with EU climate goals. Such labels are significant given that the European Union intends to mobilize at least €1 trillion in public and private sustainable investments over the next decade.⁴⁹

Many member states were staunchly opposed to nuclear power's inclusion in the green taxonomy despite its zero-emissions profile. Austria and Luxembourg threatened legal action to block the commission's proposal to include nuclear power.⁵⁰ The German governing coalition was fractured on the issue: the Greens pushed heavily to reject nuclear power and sought to follow Austria's example by also filing suit.⁵¹ The antinuclear position was opposed by France and other states heavily reliant on nuclear power. Meanwhile, Germany was keen to include natural gas in the taxonomy given the heavy role it plays in its energy system at present and over the long term as a balancing resource in a renewables-heavy grid.

This debate revealed that much of Europe, with German policymakers at the head of this coalition, did not value the ability of nuclear power to provide firm (nonintermittent) carbon-free electricity in place of natural gas-fired generation. It was believed that a renewables-heavy mix paired with natural gas was the optimal system design for Europe and did not consider the security and economic risks of reliance on Russia to be significant.

In July 2022, well into the energy crisis, the European Parliament backed the inclusion of investments in gas and nuclear power plants in the green taxonomy.⁵² In the case of gas, this inclusion was predicated on quite strict emissions intensity standards and low- or no-carbon fuel (e.g., hydrogen) readiness. The move was widely seen as an attempt to placate both Germany's attachment to natural gas as a transition technology and France's view that nuclear should be classified as a green technology.⁵³ In the end, no legal moves were taken to block this settlement; only a minority of countries, including Austria, Germany, Luxembourg, and Spain, voiced criticism at the time.⁵⁴

The green taxonomy debate highlights the deep division in Europe over the role of nuclear power despite its emissions-free profile.⁵⁵ In the years preceding the energy crisis, many member states laid out plans to phase out or reduce the share of nuclear energy in their electricity mix, with some even imposing bans on new nuclear generation. Broadly, precrisis Europe saw a limited role for nuclear power in the energy transition. Most notable, of course, was Germany's long-running commitment to eliminate all nuclear power.⁵⁶ Even France, the continent's leader on nuclear energy, had been considering a cap on the contribution of nuclear energy to its electricity supply, a policy that would lead to early plant retirements and effectively stunt the build-out of new reactors.⁵⁷

Coal-fired power, once the starting point for European integration, was broadly expected to be phased out of the European energy system. The Green Deal's renewables deployment targets, coupled with the bloc's price on carbon and tightening pollution regulations, have exerted economic pressure on coal power plants, forcing many to shut down. At the EU level, there are no direct mandates to phase out coal consumption, although many member states have established their own timelines.

As of June 2024, a total of 23 countries had announced their intent to phase out coal–with four of them (Austria, Belgium, Portugal, and Sweden) achieving the phaseout.⁵⁸ France expected to see all of its coal plants close by the end of 2022, barring a loophole that would have allowed coal plants to burn a mix of coal and biomass to stay online.⁵⁹ While 12 countries aimed to be coal-free before 2030, another seven countries had set their goals to phase out coal after 2030. The least ambitious targets were set by Bulgaria, which aimed to exit coal between 2038 and 2040, and Germany. Berlin's goal of a coal exit by 2038 has been heavily criticized as "too weak" and deemed incompatible with the Paris Agreement by German civil society groups.⁶⁰ In July 2021, Germany's supreme court ruled that the government would have to strengthen its 2030 climate targets– including its coal exit–by the end of 2022.⁶¹ A few months after the ruling, the ascendant ruling coalition released a proposal to bring forward Germany's coal exit to 2030.⁶² At the other end of the spectrum, some European countries (Bosnia-Herzegovina, Kosovo, Poland, Serbia, and Turkey) have not entertained a coal phaseout target at all.

The Crisis

Fall 2021 through Winter 2022-23

Preinvasion Gas Market Disruptions

The European energy crisis began in August 2021 with the first signs of Russian manipulation of natural gas. In August, Gazprom dramatically cut gas flows through the Yamal pipeline, which runs through Belarus and Poland, ostensibly due to a fire at a gas processing facility in western Siberia.⁶³ Through 2021, Gazprom's deliveries of natural gas remained at minimum levels despite the recovery of both gas demand and prices in the European Union following the economic downturn caused by the Covid-19 pandemic. In another example of targeted supply restrictions, Gazprom booked only 4 percent of the additional guaranteed transit capacity Ukraine offered for September. With a return of healthy demand for gas, these supply limitations sent prices surging across European gas markets.⁶⁴

The structure of its long-term contracts and its traditional role as a low-cost swing supplier allowed Gazprom to effectively manipulate the European gas market. Gazprom contracts were usually governed by a take-or-pay clause, which ensured very high compensation to the seller from the buyer if gas deliveries were not accepted as set out in the long-term agreement. These contracts offered both buyer and supplier a degree of discretion in the total volumes to be delivered, allowing a percentage of gas supplies to be adjusted according to circumstances, mainly demand and price outlooks.⁶⁵ As the Covid-19 crisis faded in 2021, Russia consistently delivered the minimum volumes allowed in the long-term contracts, a legal maneuver that reduced supplies throughout the European Union and forced procurement from higher-priced import sources.

The throttling of gas flows into Europe through 2021 left analysts perplexed. A CNBC story from August 2021 ran with the headline "Russia Is Pumping A Lot Less Natural Gas to Europe All of

a Sudden–and It Is Not Clear Why."⁶⁶ Tom Marzec-Manser, the lead European gas analyst at Independent Commodity Intelligence Services, hypothesized that Russia was applying pressure on Europe to certify the Nord Stream 2 pipeline: "Gazprom is readying itself for starting Nord Stream 2 and it is hoping to exert an element of leverage in terms of trying to make sure that when all the regulatory t's get crossed and i's get dotted, that that process is as swift as possible. . . . If there is less gas around than normal and the price is high then it may streamline that process."⁶⁷

The German government also saw little reason for concern. A now declassified October 2021 top-secret security assessment of Nord Stream 2 by the German government found little risk of certifying the pipeline: "Overall, the assessment comes to the conclusion that the granting of certification [for Nord Stream 2] does not jeopardize the security of gas supply in Germany and the European Union."⁶⁸ It added that "for both the German and neighboring markets, the risk of serious impairments to supply security due to the failure of individual import supply infrastructures is very limited." When assessing the chance that flows from the pipeline could give Russia leverage, it found that "gas deliveries from the Soviet Union and later from Russia to Germany have been reliable and contractual for decades, even in times of political tension."⁶⁹

Figure 4: Natural Gas Prices across U.S., EU, and Asian Reference Indexes, 2016-24



Source: "IMF Primary Commodity Prices," International Monetary Fund, https://www.imf.org/en/Research/commodity-prices.

Gazprom initially claimed that the reduction in deliveries was due to technical reasons associated with drilling well closures sourced from the period of lower demand during the Covid-19 pandemic.⁷⁰ However, the systematic emptying of Gazprom-controlled gas storage facilities in Europe was, in retrospect, likely part of a geo-economic coercion strategy to prepare for the invasion of Ukraine. By exporting the minimum quantity of gas permitted by long-term contracts and suspending sales on the spot market, Russia placed the European market under significant duress, resulting in record-high pricing in August 2021. Furthermore, the natural gas price differential between the U.S. Henry Hub index and the European TTF index had already reached record highs (see Figure 4).

In November 2021, U.S. intelligence agencies sounded the alarm that a Russian invasion of Ukraine could be imminent.⁷¹ Russia had moved significant forces and sophisticated weaponry near its border with Ukraine. American analysts cited by the New York Times suggested that Putin saw the coming months as a unique window of opportunity, with rising gas prices making Europe more dependent on cheap Russian supplies and Merkel's imminent departure from the chancellorship limiting the future pressure on Ukraine to make concessions.

Russia Invades

Russia's invasion of Ukraine in February 2022 initiated the worst security crisis in Europe since World War II and indicated that the unusual gas market developments of 2021 were not merely due to commercial or engineering causes. The invasion caused a major economic and political upheaval in European capitals and elevated the security of Europe's energy supply to the top of the political agenda.

Just days after the invasion, German chancellor Olaf Scholz gave his now famous Zeitenwende speech, which not only called for a massive increase in German military spending but also suspended the certification of the Nord Stream 2 pipelines.⁷² Additionally, the United States and European Union responded quickly with surprisingly robust sanctions, which included the dramatic step of freezing Russian assets in the global financial system.⁷³ In June 2022, the Russian Elites, Proxies, and Oligarchs Task Force, comprising top finance and justice officials from Australia, Canada, France, Germany, Italy, Japan, the United Kingdom, the United States, and the European Commission, announced it had blocked \$30 billion in assets belonging to Russian oligarchs and officials and immobilized \$300 billion owned by the Russian central bank.⁷⁴ Notably, no sanctions on Russian energy imports were included. Furthermore, Russian gas continued to flow into Europe via Nord Stream 1 and other pipelines.

Despite the absence of energy market sanctions, the market feared that sanctions would come, and forward gas and power prices soared on March 7. Seeking to ease market impacts, European leaders quickly downplayed the likelihood of such a sanctions package.⁷⁵ Dutch prime minister Mark Rutte noted that cutting off Russian oil and gas supplies would be unsustainable for EU member states "because we need the supply and that is the uncomfortable truth."⁷⁶ Chancellor Scholz also vigorously pushed back on calls to ban imports of Russian gas and oil.⁷⁷ While Berlin agreed to halt the Nord Stream 2 project following Russia's invasion of Ukraine, Scholz emphasized that importing

energy from Russia was an economic necessity. In a statement, the German leader noted, "At the moment, Europe's supply of energy for heat generation, mobility, power supply and industry cannot be secured in any other way. It is therefore of essential importance for the provision of public services and the daily lives of our citizens."⁷⁸

The Social Democratic Party's junior coalition partner, the Green Party, supported Scholz's position. In other words, fears of widespread economic calamity among many of Europe's most powerful countries resulted in a cautious approach.⁷⁹

One quickly emergent issue for European policymakers was the difficulty in correctly assessing the impact of the crisis due to a lack of Europe-wide data. As Bruegel found, "Technical questions on the EU's ability to survive without Russian natural gas and oil were clouded by an absence of data on the volumes and routes through which Russian fossil fuels were entering the EU."⁸⁰ European businesses were also sounding the alarm. A report from the Kiel Institute predicted a "sharp recession" if Germany were to cut itself off from Russian gas.⁸¹ German businesses reacted with horror at the prospect of a potential European embargo on Russian fossil fuels. At the end of March 2022, the head of German chemicals giant BASF asked rhetorically, "Do we knowingly want to destroy our entire economy?" There was a prevailing sense that the German economy might suffer a dramatic economic contraction.⁸²

Several economists disputed these concerns in a prominent paper produced in the weeks after the invasion.⁸³ They concluded that ending European purchases of Russian energy would entail "substantial but manageable" economic cost for the German economy. This assessment was prescient, as Russian pipeline gas flows would dramatically decline that summer due to sabotage of the Nord Stream pipelines. In a retrospective analysis, two of the original authors, Benjamin Moll and Moritz Schularick, and Bruegel senior fellow Georg Zachmann concluded, "The German economy successfully adapted to the cut-off of natural gas from Russia last year, suggesting that Germany could have moved more swiftly to sanction Russia after it invaded Ukraine."⁸⁴

The United States was also wary of energy sanctions. A total embargo on Russian oil exports would likely lead to a surge in global oil prices, drive U.S. domestic gasoline prices up, and accelerate the inflationary pressure hampering the global economy. There was also clear concern among certain Central and Eastern European states, whose energy infrastructure was less integrated with their European neighbors and therefore more dependent on Russian energy. For example, when EU import bans on Russian coal, seaborne crude oil shipments, and refined oil products eventually went into effect in August 2022, December 2022, and February 2023, Bulgaria, the Czech Republic, Hungary, and Slovakia secured exemptions to certain oil products.⁸⁵ Without these exemptions, reaching consensus at the European Union, which is required for sanctions, would have been difficult.

However, the European Union ended up passing two sanctions packages on February 23 and February 25, 2022, which focused on limiting technology transfers and disrupting the Russian financial system but did not directly target imports of Russian fossil fuels.⁸⁶ The rationale behind

these sanctions was to limit Russia's access to funds derived from hydrocarbon sales while maintaining its supply to the European Union in the short term.

In response, the Kremlin announced in March 2022 that hostile countries would have to pay for gas in rubles in a two-phase payment system (euros and rubles) using Gazprombank.⁸⁷ The European Commission opted for flexibility, allowing European companies to choose between continuing with their long-term contracts using this new mechanism or unilaterally rescinding them. As a result, many European utilities suspended their contracts, provoking a decrease in gas flows from Russia. In particular, Poland and Bulgaria refused to pay for imports in rubles, leading to the closure of the Yamal gas pipeline running through Poland to Germany and ceasing Russian exports of gas to Bulgaria, which was particularly dependent on Russian supplies. The Bulgarian economy was especially exposed to the energy shock given its heavy reliance on fossil fuels–specifically, coal for electricity generation and crude oil products for transportation.⁸⁸ After the state-owned domestic gas supplier Bulgargaz refused to accept a rubles-based payment scheme, Gazprom unilaterally suspended deliveries in April 2022.

Through the summer of 2022, Gazprom increased its pressure on the EU gas market, intermittently reducing flows through the Nord Stream 1 pipeline. Again, Gazprom gave cover explanations, this time stating that technical disruptions to key equipment repairs were due to Western sanctions.⁸⁹ By June, surging energy prices (Figure 4) were beginning to cause a cost-of-living crisis affecting people and businesses across the European Union. The economic impacts placed political pressure on European leaders. In a televised interview marking Bastille Day, French president Emmanuel Macron warned his constituents to prepare for a complete cutoff of Russian natural gas, advanced initiatives to diversify the French energy mix, and spoke of a national "sobriety plan" to conserve energy.⁹⁰ German economy and climate minister Robert Habeck said in June 2022, "It's a tense, serious situation... It must be acknowledged that Putin is reducing the gas supply to Europe bit by bit, also to drive up the price, and we must respond with our measures."⁹¹

NORD STREAM SABOTAGE

On September 26, 2022, two underwater explosions damaged three of the four pipelines that constituted the Nord Stream 1 and 2 pipelines off the Danish island of Bornholm (the latter was completed in 2021 but was not in operation). The sabotage, whose perpetrators have not yet been identified, left the infrastructure inoperable, closing the main gas supply line between Russia and the European Union.

Despite the lack of evidence warranting conclusive attribution, two schools of thought have emerged pointing to either Russian- or Ukrainian-backed groups as the perpetrators. The circumstantial evidence in the immediate aftermath pointed toward the Kremlin.⁹² Russia's Baltic fleet, capable of deploying divers and miniature submarines, is based in Kaliningrad, just 300 kilometers from the sabotaged pipelines. The incident occurred a day before a new pipeline was completed to connect Poland with Norwegian gas via Denmark in an effort to reduce Warsaw's reliance on Russian gas. Another narrative that subsequently gained traction centers on Ukrainian involvement. Ukraine has carried out a number of dramatic intelligence operations since the war began, some of which have roiled Ukraine's backers, such as the assassination of Russian television commentator Daria Dugina. Kyiv, which long opposed the pipelines, may have feared Russia could successfully blackmail Germany and others to reduce their support for Ukraine. In March 2023, German media carried a report that German police had traced a yacht thought to have been used to transport explosives to the sabotage site to two Ukrainian nationals.⁹³ These reports aligned with New York Times reporting that U.S. intelligence officials suspected involvement by renegade pro-Ukrainian groups.⁹⁴ Although German authorities are still investigating the explosions, Denmark's and Sweden's public prosecutors closed their inquiries into the explosions in February 2024.⁹⁵

Whoever was responsible, the attack eliminated Russian gas flows to Germany, led to sharp increases in energy prices, and firmly demonstrated that Europe's energy links to Russia could not be isolated from the broader geopolitical conflict.



Figure 5: Russian Gas Output by Pipeline Infrastructure, 2021-23

Source: "European natural gas imports," Bruegel, June 2024, https://www.bruegel.org/dataset/european-natural-gas-imports.

The Perfect Storm: Overlapping Crises in Electricity Markets

The crisis in the European natural gas sector intersected with two incidental factors–a severe continental drought and the poor performance of the French nuclear fleet–to cause a matching crisis in the European electricity sector. Electricity markets saw prices reach historic levels, which devastated consumers, destabilized energy companies, and raised serious resource adequacy concerns. Heading into the winter of 2022-23, system planners and political leadership publicly warned of the risk of load shedding, a process by which consumers are intermittently disconnected to preserve overall system stability.

DROUGHT

In 2022, severe drought, characterized as the worst in 500 years, simultaneously affected the Mediterranean basin, Central Europe, and the Nordic countries.⁹⁶ Lack of rainfall severely reduced the generation output of many hydroelectric dams, which represent roughly 16 percent of total European generation capacity. In 2021, these generation assets produced 805 terawatt-hours (TWh) of electricity, representing 17 percent of all production; only gas-fired generation (26 percent) and nuclear-fired generation (23 percent) contributed a larger share. Drought conditions in 2022 caused hydroelectric generation to fall 15 percent year over year to only 687 TWh, the lowest level since 2011, despite the very strong financial incentive of record-breaking spot prices to produce at maximum capacity.

Drought further affected the electricity sector by reducing water levels in European rivers. The conditions restricted navigability by barge and, in some cases, interrupted delivery of fuel to coal-fired power plants on the Rhine, taking additional generation capacity out of the system.⁹⁷ In France, low water levels caused restrictions on the discharge of cooling water by nuclear power plants, causing Électricité de France (EDF), the French nuclear operator, to take plants offline or reduce output. Eventually, this problem was partially mitigated when the relevant environmental agency granted waivers at the government's behest on the basis of public necessity.⁹⁸

FRENCH NUCLEAR OUTAGES

France possesses the largest nuclear power fleet in Europe by a wide margin: 61 gigawatts (GW) of production capacity spread out over 56 reactors, dwarfing the next-largest fleets in Ukraine (14 GW) and the United Kingdom (8 GW).⁹⁹ This fleet typically supplies over 70 percent of France's electricity.¹⁰⁰

As gas and electric prices surged through early 2022, the French nuclear fleet was rocked by the discovery of widespread stress corrosion cracking.¹⁰¹ Problems similar to those initially identified at a single plant were soon discovered at numerous others, forcing EDF to take several plants offline. These events coincided with prescheduled and unavoidable 10-year maintenance outages. In total, an unprecedented number of nuclear reactors were taken offline just when the French and broader European electric grid needed them most. The nadir of French nuclear output occurred in July-August 2022, when only 18 TWh was generated out of a possible 45 TWh, a 40 percent capacity utilization rate. On an annual basis, 2022 was the worst performance in decades, with output falling over 22 percent versus 2021.



Figure 6: French Nuclear Fleet Electricity Generation and Capacity, 2010–23

Source: "Ember Electricity Data Explorer," Ember Climate, June 2024, https://ember-climate.org/topics/nuclear/.

Although France typically exported electricity to its neighbors, the combination of drought and nuclear outages made it a net importer of electricity for the first time in 42 years. Rather than support supply deficits in neighboring nations such as Germany, Spain, and the United Kingdom, it drew on power from these nations, exacerbating supply shortages and high prices in those countries. However, this result demonstrated the value of an integrated electricity system linked both physically with transmission capacity and commercially with liquid and dynamic market mechanisms. Such integration allows the sharing of generation resources, which improves overall system reliability and brings down the cost of achieving that reliability. Without this integration, France would likely have been forced into economically and socially painful cuts to electricity in the summer of 2022 and winter of 2022-23.

HISTORIC PRICE VOLATILITY

Prices in the European electricity market reached record highs through 2022. Drought and nuclear outages knocked low-priced baseload capacity out of the supply stack. The crisis in the gas market meant gas-fired generation capacity struggled to secure fuel. Plants that could secure fuel did so at exorbitant prices, which meant that the electricity they produced was offered on electricity markets

at a steep premium. Additionally, carbon dioxide prices in the EU Emissions Trading System market traded to historic highs, reaching a peak of €100 per ton equivalent in February 2023.¹⁰² Higher carbon dioxide costs borne by fossil-fired generation (i.e., coal and gas) likewise translated into higher prices offered in electricity markets.



Figure 7: France Becomes Net Importer of Electricity between 2021 and 2022

Lines and values represent net imports in electricity between France and each

Source: Dave Jones et al., *European Electricity Review 2023* (London: Ember, January 2023), https://ember-climate.org/insights/ research/european-electricity-review-2023/#supporting-material. Reproduced under CC-BY-SA 4.0.

The structure of electricity markets is crucial for understanding how these factors translate into higher electricity prices. Electricity markets are settled on the margin: Generators offer their electricity into the market and are dispatched to meet demand in order of lowest to highest offer price, a process called merit order dispatch. The offer price of the last unit dispatched, the marginal unit, is the price received by all generators (supply) and the price paid by all buyers (demand). This price formation model is used in all restructured electricity markets around the world, most notably in U.S. electricity markets.

Europe has made substantial progress on deploying renewables that produce electricity at zero marginal cost and are therefore first in merit order dispatch. However, demand levels almost always reach beyond the generation output of renewables, causing merit order dispatch to move deeper into the supply stack and dispatch fossil-fired generation to balance supply with demand. Renewables are therefore rarely the marginal unit and therefore rarely set the market-clearing spot

price. Nonetheless, they play a key role in reducing overall system costs by reducing how often and deep the market must go into the supply of high-priced fossil-fired generation to reach equilibrium.

With large volumes of low-cost hydropower (first to be dispatched in the supply stack) and nuclear power out of the supply mix, gas-fired generation units priced at steep premiums set the margin in almost every country. Through this mechanism, the historically high price of natural gas translated directly into historically high prices for electricity.¹⁰³



Figure 8: Electricity Prices in Select EU Markets, 2020-23

Source: "European power price tracker," Ember Climate, June 2024, https://ember-climate.org/data/data-tools/europe-power-prices/.

In the years before the crisis, average annual wholesale prices for most European countries were generally €30-€50 per megawatt-hour (MWh). By early 2022, electricity prices, tracking the movement in gas prices, had reached monthly averages of over €200 per MWh in most countries. Pricing surged higher through the summer as the combined effect of nuclear and hydroelectric underperformance drove increasing reliance on extremely high-priced gas generation. Increased power burn–or gas consumed by power plants to produce electricity–intensified pricing pressure in the gas market as buyers sought to fill gas storage facilities for the upcoming winter. This drove gas prices, and by extension electricity prices, ever higher through the summer. Peak pricing

occurred in August, with most countries averaging well over €400 per MWh. Italy fared the worst, with a monthly average of €543 per MWh.

Yearly averages for 2022 for most countries settled at triple the typical annual average of the recent past or more. The impact of such pricing on households, businesses, and the overall economy is enormous. A rough estimate serves to demonstrate the scale: In 2022, Germany consumed 540 TWh of electricity priced at an annual average price of €237 per MWh, representing an annual spend of €128 billion. Priced against the preceding five-year average of €35 per MWh, the same demand results in an annual spend of only €19 billion. Despite the introduction of various subsidy mechanisms, Greece's electricity prices also spiked by 40 percent for industries and 24 percent for households in 2022.





Source: "European wholesale electricity price data," Ember Climate, June 2024, https://ember-climate.org/data-catalogue/european-wholesale-electricity-price-data/.

The magnitude of extreme pricing was not uniform. Nordic countries fared better because their power mixes relied less heavily on natural gas in favor of hydro and nuclear generation. Poland likewise has relatively low reliance on natural gas in favor of heavy reliance on coal-fired power. Portugal and Spain were notable exceptions because they introduced a gas price-capping mechanism, known as the Iberian exception, which the European Commission approved after tough discussions. This gas price cap limited the price of natural gas used for electricity generation to €40 per MWh, with the government compensating gas-fired power plants for the difference, thus successfully lowering overall

electricity prices for consumers during 2022, despite indirectly increasing Spain's gas consumption for power generation and, especially, electricity exports to France.¹⁰⁴

European electricity demand historically peaks through the winter months. The winter of 2022-23 posed a worrisome risk for the overall reliability of the electric power system, and officials within the European Union, national governments, and grid operators all issued warnings on the risk of load shedding or, far worse, uncontrolled blackouts.¹⁰⁵ One area where Europe was fortunate is that both the winters of 2022-23 and 2023-24 were relatively mild, enabling lower energy usage.¹⁰⁶



Figure 10: Changes in EU Electricity Supply and Demand Balance, 2021-23

Source: "European Union," Ember Climate, May 2024, https://ember-climate.org/countries-and-regions/regions/european-union/.

The Response

Policy Interventions and Policy Costs

Energy Market Interventions

Imports of Russian gas into the European Union fell precipitously from 150 bcm in 2021 to 78 bcm in 2022 and then again to 42 bcm in 2023.¹⁰⁷ To enable and respond to these changes in fuel imports, EU member states took a wide range of steps to enhance domestic energy supply capacities and seek alternatives to Russian imports.

EUROPEAN GOVERNMENTS BRIEFLY TURN TO OLD COAL POWER PLANTS

The crisis led to a temporary reactivation of coal-fired power plants to provide resource diversity and additional capacity given that gas combined-cycle plants were heavily fuel constrained. According to a report by Ember, an estimated 13.5 GW of coal-fired plants were reactivated from recent retirement or otherwise placed on standby in supply reserve facilities, increasing the EU coal fleet by 12 percent.¹⁰⁸ These units contributed to increased coal burning through the crisis. For the first time since 2015, coal-fired generation output increased year over year in 2021 and again in 2022, reaching 448 TWh compared to 352 TWh of output in 2020.

Ultimately, due to fortuitous winter weather outcomes and effective policy intervention at the EU and national levels in other areas, these reactivated plants saw relatively marginal utilization. By the spring of 2023, the European Union had experienced an overstock of coal in its ports, which was resold to Asia and other emerging markets.¹⁰⁹ In 2023, coal-fired power generation fell relative to 2022 and returned to its long-term pattern of decline and a new EU-era low of 335 TWh.

EUROPEAN GOVERNMENTS SEEK TO PRESERVE THEIR NUCLEAR POWER GENERATION

As with coal generation, the energy crisis forced policymakers to revisit nuclear power policy. In the short term, postponing imminent reactor shutdowns was an option for policymakers. In January 2023, Belgium opted to extend the life of two nuclear reactors slated to close in 2025 by 10 years, until 2035.¹¹⁰ In Germany, considerable debate over the status of three recently shuttered nuclear reactors and three others in operation but scheduled for closure did not ultimately lead to policy changes.¹¹¹ In April 2023, Germany took its last three remaining nuclear power plants offline.

Crisis impacts on the nuclear power sector will likely play out over the long term as countries plan investments that shape the overall direction of their generation resource mix. French nuclear policy is demonstrative. Precrisis plans proposed phasing down nuclear power's contribution to the overall electricity mix from over 70 percent to below 50 percent–a policy that effectively eliminates new plant builds.¹¹² Postcrisis French policy has inverted the 50 percent target from an upward limit to a minimum threshold, and the government has announced plans for a new reactor-building campaign.¹¹³ France has also organized a coalition of European countries interested in pursuing expanded nuclear power generation.¹¹⁴

Overall, nuclear power accounts for about a quarter of European energy, with about half of that coming from France.¹¹⁵ At present, France and Slovakia are the only European countries with nuclear reactors currently under construction. In March, Belgium hosted a Nuclear Energy Summit with 30 countries, a sign of the newfound political interest in nuclear power. Poland signed a \$40 billion agreement with U.S. companies to construct two nuclear power plants, with three reactors each, the first opening in 2033.¹¹⁶ Polish and South Korean companies have also agreed to build additional nuclear plants in Poland.¹¹⁷ The Czech Republic is advancing plans to construct multiple new reactors for delivery in the mid-2030s.¹¹⁸

NORWEGIAN GAS STEPS UP

In 2022, Norwegian piped gas flows into the European Union increased about 8 percent year over year.¹¹⁹ In 2023, Norway provided 109 bcm of natural gas from the Norwegian continental shelf to Europe.¹²⁰ In December 2023, Norwegian gas deliveries provided 11.1 bcm (121.7 TWh) in one month, the largest amount ever provided to Europe in a single month.¹²¹ Norway now provides approximately 30 percent of EU natural gas and has become the largest EU gas supplier, with the United States second at approximately 19 percent. Norway has also replaced Russia's leading role in providing gas to Germany.¹²² In 2021, Norway provided under 20 percent of Germany's natural gas; in 2023, that figure stood at 48 percent.¹²³

For the European Union, Norway is a far more secure partner. Although Norway is not a member of the European Union, it is part of the European single market. Further, Norway is a founding member of the North Atlantic Treaty Organization (NATO) and is deeply embedded in the transatlantic efforts to counter Russian aggression. Substituting Russia for Norway as Europe's dominant supplier indicates a significant improvement in European energy security.
The precrisis preference for Russian gas was based on price. Thus, the improvement in energy security comes at a cost, as Norwegian gas is structurally more expensive. The new equilibrium is therefore not without its detractors. Some European commentators have accused Norway of profiting off of the war and Europe's economic pain.¹²⁴ Europe's newfound dependence on Norway and the Norwegian energy giant Equinor has created some market concern that any disruption, such as from maintenance needs, would lead to price spikes. As a Bloomberg report surmises, "So long as the bloc continues to depend heavily on fossil fuels, Norwegian hydrocarbons will be essential to keeping the lights on in Europe."¹²⁵ Other suppliers of piped gas, such as Algeria and Azerbaijan, have not seen a comparably significant increase in supply to Europe.



Figure 11: EU Pipeline Gas Imports, 2021-24

Source: "European natural gas imports," Bruegel, June 2024, https://www.bruegel.org/dataset/european-natural-gas-imports.

Most surprising is a lack of concerted effort to increase Europe's domestic gas production. Despite the ongoing crisis of gas supply and price, domestic gas production within the European Union has continued its downward trajectory. Between 2016 and 2023, EU gas production more than halved, primarily due to the natural depletion of fields and challenges related to political and social acceptance.¹²⁶ Since the invasion of Ukraine, natural gas production in the European Union

has experienced a significant decline of approximately 13 percent due to the May 2024 closure of the Dutch gas field in Groningen, the bloc's major gas source. One positive indicator for supply is in Romania, which foresees a rise in gas production thanks to upcoming projects in the Black Sea.¹²⁷ Denmark's gas output is also expected to receive a boost with the delayed return of the Tyra field in 2024.¹²⁸

EUROPE TURNS TO LNG, PARTICULARLY FROM THE UNITED STATES

The primary focus of transatlantic cooperation has been securing LNG volumes for Europe. A March 2022 U.S.-EU announcement featured prominent commitments by the United States to expand LNG exports to Europe to at least 50 bcm per year by 2030. The agreement started with a goal of achieving 15 bcm of increased exports in 2022, a target that was swiftly met and surpassed by September of that year.



Figure 12: U.S. LNG Exports to Europe, 2016-23

Source: "U.S. Natural Gas Exports and Re-Exports by Country," U.S. Energy Information Administration, June 2024, https://www.eia.gov/dnav/ng/ng_move_expc_s1_a.htm.

The surge in U.S. LNG exports to Europe is stunning. In 2021, total U.S. LNG exports to Europe stood at 29 bcm; 2022 saw exports more than double to 70 bcm. These flows played a critical role for Europe as it sought to replace lost Russian pipeline volumes. In total, U.S. LNG grew to 15

percent of all EU gas imports in 2022.¹²⁹ Policymakers and officials on both sides of the Atlantic have touted this result as a demonstration of the ongoing strategic partnership between Europe and the United States. In June 2022, in a joint statement on European energy security, President Biden and European Commission president Ursula von der Leyen said, "While Russia has cut supplies of natural gas to several EU Member States, the United States and other producers have stepped up."¹³⁰

In remarks at the North American Gas Forum in October 2023, U.S. assistant secretary for energy resources Geoffrey R. Pyatt noted,

As I travel around the world, especially in Europe, the message I always hear is hurray for American LNG producers. There is great appreciation for the role that American producers have played, especially in this period of Russia-caused disruption. From a strategic and foreign policy standpoint it's a good thing that our export capacity, our liquefaction capacity is going to grow as much as it is over the next couple of years as we look out towards 2025, 2026.¹³¹

The lines of communication and coordination afforded by the task force were crucial in enabling ad hoc diplomatic efforts on energy matters. A notable example of this occurred in February 2022 when Japan agreed to divert LNG cargoes to Europe following determined diplomatic engagement with U.S. and EU officials.¹³² However, much of the result was independent of direct government action. The United States has no institutional mechanisms by which to direct LNG exports to specific markets, countries, or customers. Instead, decisions on long-term investment, production, and export destinations are market-informed and taken by private companies.

The reduction and then abrupt cutoff in Russian pipeline volumes were, in effect, an exogenous, outward shift in the LNG demand curve and have resulted in a new, far higher equilibrium point for European LNG prices. This created the price signal, as reflected by the historic highs of the TTF European gas price index, for firms to redirect LNG shipments to the European market.

The U.S. LNG export industry possesses unique flexibility to respond to price relative to the rest of the global LNG market. Most U.S. export volumes are sold through a free-on-board contract model that gives buyers ownership of cargoes at the export terminal and offers destination flexibility.¹³³ This leaves buyers free to send cargo wherever they wish based on commercial considerations– typically the market that can command the highest price.

These contracting structures have facilitated a shift from long-term point-to-point contracts toward a "portfolio player" model in which firms buy up a large pool of contracts for their own procurement needs and resell to whatever market commands the highest premiums.¹³⁴ Portfolio players make up a growing share of U.S. supply deals and have played a key role in shifting LNG volumes to Europe.¹³⁵

Fortuitous market conditions have also created spare capacity and have made redirection of LNG volumes to Europe feasible. In China, slowed economic growth and Covid-19-related restrictions reduced energy demand, driving LNG imports 20 percent lower year over year and thereby freeing up over 20 bcm of LNG for other destinations.¹³⁶



Figure 13: U.S. Export Destinations by Region, 2018-23

Source: "U.S. Natural Gas Exports and Re-Exports by Country," U.S. Energy Information Administration, June 2024, https://www.eia.gov/dnav/ng/ng_move_expc_s1_a.htm.

The dramatic shift of volumes away from Asia to Europe highlights how the European energy crisis cascaded effects globally. From the global perspective, successful transatlantic partnerships to redirect LNG volumes are viewed less positively. The primary Asian gas price index, JKM, effectively traded at parity with the TTF European gas price index throughout the energy crisis (see Figure 4). Many Asian buyers of LNG, both private and government buyers, were exposed to price volatility and energy insecurity sourced ultimately from the European crisis.¹³⁷ Lower-income countries such as Pakistan suffered from energy shortages even as Europe largely succeeded in averting such shortages.¹³⁸ These outcomes undoubtedly have consequences for future energy strategy and wider geopolitical developments in the region.

LNG Infrastructure Construction

One of the most relevant measures to substitute Russian gas is the construction of new LNG terminals. The majority of these projects have involved the rapid and cost-effective construction of new floating storage and regasification units (FSRUs), totaling 16 projects, in addition to the expansions of 11 existing LNG terminals.¹³⁹ Germany did not have any LNG terminals when the war

erupted; it pushed through FSRU permitting and "onshore connections in record time," with the first FSRU coming online in December 2022.¹⁴⁰





Source: "European LNG Tracker," Institute for Energy Economics and Financial Analysis, February 2024, https://ieefa.org/european-Ing-tracker.

This rise in LNG imports positioned the European Union as the world's largest LNG importer in 2023, ahead of other traditionally bigger importers such as China and Japan.¹⁴¹ One of the fears of this rapid development of LNG terminals is the risk of building overcapacity that is eventually not used, as happened on the Iberian Peninsula in recent decades.¹⁴² The disparity between Europe's LNG capacity and demand is widening in a context of significant infrastructure expansions and LNG imports plateauing since late 2022.¹⁴³ The utilization rate of Europe's LNG terminals averaged 58 percent in early 2023, prompting questions about the necessity of further infrastructure expansion, projected until 2027.¹⁴⁴ However, as noted by the EU Agency for the Cooperation of Energy Regulators, around 80 percent of capacity additions between 2022 and 2024 were FSRUs, which allow for the repurposing or relocation of floating infrastructure at a reasonable cost.¹⁴⁵

Additionally, there are plans to establish four onshore conventional plants, with construction timelines scheduled between 2023 and 2027. Germany has been the main player in this expansion following the approval in record time of the LNG Acceleration Act, which involves the federal government in infrastructure development.¹⁴⁶ This approach has marked a significant departure from prior policies, with the state actively engaging in project implementation, including direct involvement in financing and operational oversight. Most operators of the new Germany LNG plants

are expected to switch to climate-neutral hydrogen by 2044 at the latest. But the exact timing, techno-economic feasibility, and adherence to these timetables remain to be seen.





Source: "Global Gas Infrastructure Tracker," Global Energy Monitor, June 2024, https://globalenergymonitor.org/projects/global-gas-infrastructure-tracker/tracker/.

Croatia, Greece, and Italy have also been very active in the construction and expansion of new LNG terminals and hope to serve as a gateway for gas to neighboring landlocked countries such as Austria, the Czech Republic, Hungary, and Slovakia. Two infrastructure initiatives stand out. First, the Vertical Corridor, presented in January 2024, will make it possible to transport natural gas from Greece and Bulgaria to Hungary, Romania, and Slovakia.¹⁴⁷ The pipeline is expected to extend to Moldova and underground storage facilities in Ukraine, increasing natural gas import and storage options for one of Gazprom's remaining export markets in Europe. Both Greek and EU subsidies

have enabled the construction of pipelines that span the country and connect to a new import terminal that will provide Central and Eastern Europe with gas for decades to come. Greece's ambition to become a key European LNG import hub has been aided by forceful lobbying efforts from U.S. officials for European nations to use Greece's new terminal and pipelines.¹⁴⁸ Second, the Adriatica pipeline will allow Italy to reconfigure its gas flows, traditionally designed to go in a north-south direction.¹⁴⁹ With this improved transmission network, Italy will be able to export gas entering from the south (i.e., from Algeria, Azerbaijan, and Libya), in addition to exporting LNG to northern neighbors, especially Austria and, subsequently, Germany.

EUROPE DRAMATICALLY CURTAILS DEMAND

Amid the significant supply-side shocks, natural gas demand in the European Union declined by 55 bcm, or 13 percent, in 2022–the steepest year-over-year fall in history. By March 2024, gas consumption in the European Union had decreased by 24 percent compared to March 2021.¹⁵⁰ There was a robust public campaign to save electricity, encouraging Europeans to turn down thermostats, take shorter showers, and turn off nonessential lighting. The lights on the Eiffel Tower, for instance, were turned off at night.¹⁵¹ These efforts had considerable impact, as EU-wide electricity demand declined by 3.4 percent in 2023 compared to 2022 and was 6.4 percent lower than its pre-crisis level in 2021.¹⁵²

The greatest decline in natural gas use was among the three main consumers: EU households to generate heat, the energy sector to generate electricity, and the industrial sector to manufacture final products such as ammonia, fuels, ceramics, and paper.¹⁵³ Although gas demand in the electricity sector temporarily increased during 2022 due to the decline in nuclear and hydroelectric power, this trend reversed in 2023 thanks to a significant increase in renewables, lower electricity consumption, and improved nuclear and hydro output. The buildings sector experienced a reduction of almost 20 percent in the winters of 2022 and 2023, attributed to milder weather, behavioral changes in response to high prices, increased fuel poverty, efficiency improvements, and record heat pump installations.¹⁵⁴

In the industrial sector, gas consumption has fallen by around 25 percent between 2021 and 2022, mainly due to reduced production and, to a lesser extent, fuel switching.¹⁵⁵ Some factors, such as price-sensitive fuel switching in the refining sector or weather effects, are expected to be temporary. Others, such as increased renewable capacity, improved efficiency, the adoption of heat pumps, and permanent industrial plant closures, are structural and lead to lasting, sometimes undesired reductions.

RENEWABLES DRAMATICALLY EXPAND AND INCREASE EUROPE'S RESILIENCE

Europe's expansion of renewables prior to the crisis helped reduce the impact of the crisis and gave Europe greater energy resilience. Renewable energy played a considerable role in meeting shortfalls in the electricity sector from lower hydroelectric and nuclear output. Renewables output increased by 80 TWh year over year in 2022 and another 87 TWh in 2023.

In 2022, wind and solar represented 22 percent of EU electricity generation, overtaking natural gas for the first time.¹⁵⁶ Solar capacity was rapidly deployed during the crisis, pushing installed capacity to 256

GW in 2023, a 56 percent increase versus 2021. Wind capacity also grew, albeit far slower than solar, even as the sector struggled with considerable supply chain and financial difficulties (Figure 16).¹⁵⁷

Figure 16: Rapid Renewables Deployment in the EU, 2018-23



Source: "Electricity Data Explorer," Ember Climate, June 2024, https://ember-climate.org/data/data-tools/data-explorer/.

The rapid deployment of renewables, primarily solar, has been fueled by record-high wholesale prices improving the financial attractiveness of renewables, by member state and commission efforts to ease permitting processes, and by the continuing fall in solar costs due to expanding Chinese solar manufacturing capacity.¹⁵⁸ The development of distributed solar, especially rooftop photovoltaic (PV) systems in households and businesses, has been one of the great social responses to the energy crisis, becoming a widespread phenomenon in the European Union.

Setting aside trade challenges, the massive expansion of solar has created new internal challenges for electricity markets. In Germany, the expansion of solar has created price cannibalization "leading to destruction of solar hour prices and solar profitability," a result that should lead to slower expansion, according to an SEB Research report.¹⁵⁹ With installed solar capacity in Germany now over 82 GW, output on a typical sunny day far exceeds the average demand of roughly 52 GW. Spain is having a similar situation, where power is at times "almost free" due to solar production.¹⁶⁰ For countries with deep solar penetration, the only conceivable path for continued expansion in the solar segment is matching investment in energy storage and transmission.

Biogas is a small, easily overlooked but significant area of renewable energy expansion. The anticipated increase in domestic biogas and biomethane production is projected to partially offset the decline in conventional gas production. Over the past decade and supported by generous subsidy packages, the European Union has witnessed a doubling in its production of biogas and biomethane, reaching approximately 19 bcm in 2022, which equates to around 6 percent of final EU gas consumption.¹⁶¹ The REPowerEU plan aims to achieve a combined volume of 35 bcm by 2030, nearly doubling current production levels.

EUROPEAN INFRASTRUCTURE AND MARKET INTEGRATION

Overall, European resilience has been greatly boosted by existing cross-border infrastructure networks and markets. Gaps in supply or infrastructure in one country can be offset by the capabilities of another–a result physically enabled by pipeline and electric transmission networks and commercially enabled by common market systems. For example, Germany did not possess any LNG import terminals until its first FSRU came online in December 2022.¹⁶² Before this, Germany was nonetheless able to access LNG supplies through shipments arriving at EU ports in countries such as the Netherlands. Europe benefited from the continued operation of infrastructure and markets because, on the whole, European countries refrained from hoarding energy, nationalizing energy supplies, or restricting outbound sales. This unity was matched on the demand side as Europeans collectively cut demand for gas and electricity in more or less similar ways.



Alexandros Paterakis, deputy CEO of Public Power Corporation S.A., speaking at the 2024 CSIS Energy Security and Geopolitics Conference.

"We had some strong reserves in the form of gas which we could capitalize on. . . . We looked at how we could, first of all, handle our own needs and then how we could play a significant role in enabling neighboring countries to get out of this situation. . . . At the same time, renewables was where we doubled down."

— Alexandros Paterakis, Deputy CEO, Public Power Corporation S.A.¹⁶³

Additionally, there was rapid movement to complete new interconnection projects, where possible, to create greater resilience. By May 2022, the Baltic states had enabled new interconnections with Poland.¹⁶⁴ By September 2023, the Baltic Pipe was completed, connecting Poland to Norwegian gas supplies through Denmark, which was crucial for improving the energy security situation of the most exposed eastern EU members.¹⁶⁵

Country-Level Fiscal and Regulatory Response

The energy crisis and the increase in wholesale energy prices in Europe prompted governments to put in place measures to shield consumers from the direct impact of rising prices, protect strategic assets, and secure the energy supply. There were fears of a deep European recession as well as concerns about inflation.

Many European governments relied on fiscal spending to keep industries afloat and alleviate the costs to the public. According to a Bruegel report, the national fiscal policy responses to the energy crisis in the European Union reached €540 billion, of which €158 billion was earmarked by Germany alone.¹⁶⁶ Separately, the United Kingdom provided €103 billion in fiscal support. The different fiscal capacities of member states within the European Union, both in terms of revenue levels and indebtedness, resulted in an overall unbalanced response that created tensions among member states.

The European Central Bank also found that the crisis contributed "to a deterioration in the current account balance of the euro area, from 2.2% of GDP in 2019 to -1.0% of GDP in 2022. Both deficit and surplus countries were affected."¹⁶⁷ This has limited Europe's fiscal capacity to invest in defense, support Ukraine, and further fund the energy transition.

Ultimately, these financial outlays demonstrate that European states had the fiscal and policymaking capacity to shield their economies from the worst effects of the crisis. As noted, both before the crisis and in the early stages, there was a widespread assumption that the loss of access to Russian natural gas would plunge the European Union into a deep recession. Some economists predicted a 10 percent decline in German output and the loss of 2-3 million jobs. Fiscal outlays, effective substitution to other energy suppliers, and demand reduction measures prevented a deep economic recession. Instead, as Moll, Schularick, and Zachmann concluded in a Brookings study in September 2023, "Overall, while the German economy is stagnating and faces substantial long-run headwinds, the direct economic costs of the end of Russian energy imports proved moderate and manageable."¹⁶⁸

Eight other EU countries were more dependent on Russian energy than Germany in 2022, including Finland, Greece, Hungary, Lithuania, the Netherlands, Poland, and Slovakia Greece, for instance, spent about 5 percent of GDP, or about €10 billion, on energy support measures, with most of the funding directed at subsidizing household energy bills.¹⁶⁹ Yet despite the price shock, Greece reduced its energy imports from Russia by more than 70 percent, much more than the EU average of 37 percent. Despite the costs and the price shock, industrial production grew by 2.3 percent in 2022, with the economy growing overall by nearly 6 percent. In the Netherlands, consumption of natural gas dropped by roughly 30 percent, but overall industrial output increased significantly.¹⁷⁰ Since the summer of 2021, European governments have implemented various policies aimed at alleviating the impact of rising prices on consumers and businesses, including temporary reductions in energy taxes for transport fuels, natural gas, and electricity; retail energy price caps; energy tariff regulations; and support programs for energy-intensive industries. Europe also experienced a wave of energy utility nationalization, including Russian-controlled companies such as Gazprom Germania (now SEFE) and those overly exposed to Russian energy supplies and market volatility and in need of a bailout, such as Uniper.¹⁷¹

For example, Germany's \in 200 billion aid package allowed large-scale consumers to apply for a gas block tariff and pay 70 percent of their gas use, calculated over the previous year's consumption, at a subsidized cost of \in 0.70 per kilowatt-hour (kWh).¹⁷² Additionally, governments made efforts to stabilize wholesale prices and ensure energy security by promoting energy savings and capping energy costs, particularly for wholesale electricity and gas. Some countries also attempted to reform energy markets, while windfall taxes targeted inframarginal power generators, crude refiners, and upstream producers in a temporary and exceptional manner.

Countries also implemented direct transfers, such as discounts at gas stations or vouchers– frequently for vulnerable households–often at high cost for a dubious return. One widespread practice, often by regional or local authorities, was to reduce public transport fares, encouraging the use of public transport instead of private vehicles as a fuel-saving measure. However, criticism arose due to the failure of many measures to specifically target vulnerable groups, instead focusing on mitigating social and political unrest.¹⁷³ The prevalence of untargeted price-distorting measures contradicted the need to incentivize energy conservation and honor existing commitments to the energy transition and to curve fossil fuel subsidies. However, despite some of these measures, demand declined significantly in Europe.

In addition to the fiscal costs, the energy crisis further fueled inflation in Europe. The European Central Bank found that "high energy and commodity prices and supply bottlenecks have driven strong inflation in the euro area and higher inflation differentials between euro area countries."¹⁷⁴ Countries with greater dependence on Russian energy, such as Latvia, suffered inflation rates above 10 percent. Unlike the United States, where inflation was mainly demand driven, the supply shock in the European Union generated by the higher energy prices and the significant import dependency was responsible for the lion's share of inflation in 2022-23.¹⁷⁵ For example, headline inflation averaged 9.3 percent in Greece in 2022, driven largely by surging energy prices.¹⁷⁶

EU Policy Response

The Russian invasion of Ukraine caught the European Union off guard. The EU institutions had largely focused on the climate dimensions of energy policy and not on the security of supply, which was seen as the responsibility of member states. Yet Russia's invasion of Ukraine demanded a collective EU response. The European Union helped ensure a unified response and set direction for its member states, from creating collective agreement on the need to decouple from Russian energy to setting targets for demand reduction and ensuring that Europe would not abandon its climate goals.

Europe demonstrated its unity in the Versailles declaration of March 11, 2022.¹⁷⁷ Although the declaration was an informal and nonbinding document, it demonstrated the political will of European leaders to take a step forward and reinforce the bloc's strategic autonomy, especially in three directions: bolstering defense capabilities, reducing energy dependencies, and building a more robust economic base. On the energy front, the declaration called on the European Commission to present REPowerEU, a plan to address the energy shock and ensure that the need to decouple from Russian energy would not undermine the climate ambitions embodied by the EU Green Deal.

REPowerEU, primarily a declaratory document of the European Commission, provides the states with the tools (e.g., budget and legislative mechanisms) to implement the proposals it contains. While some of the measures and targets set out in REPowerEU have subsequently been incorporated into the EU acquis (EU law), such as the 2030 renewable energy targets, others have been ignored or rendered obsolete, such as those related to imported renewable hydrogen. The REPowerEU budget was set at approximately €300 billion, with around 75 percent funded through loans from the Recovery and Resilience Facility (NextGenerationEU funds) and the remaining 25 percent coming from grants funded by the auctioning of additional Emissions Trading System allowances and other EU sources. Member states could access this funding by including REPowerEU measures in their national recovery and resilience plans, which the European Commission then evaluated.

REPowerEU outlined an ambitious strategy aimed at accelerating the deployment of renewable energy across power generation, industry, buildings, and transport sectors, and proposed raising the 2030 renewable energy target set under the Fit for 55 packages from 40 percent to 45 percent.¹⁷⁸ These included a dedicated EU solar strategy to double solar PV capacity by 2025 and reach 600 GW by 2030, alongside the Solar Rooftops Initiative mandating installation of solar panels on new public, commercial, and residential buildings.¹⁷⁹ REPowerEU also highlighted efforts to double the deployment rate of heat pumps, integrate geothermal and solar thermal energy into modernized heating systems, and streamline the permitting process for major renewable projects. Additionally, the commission aimed to establish a target of 20 million tons of renewable hydrogen by 2030, half of it to be imported, and launched a Biomethane Action Plan to boost production to 35 bcm by the same date.

REPowerEU also catalyzed a legislative proposal to establish the minimum gas storage mandate. This storage mandate set the target for 80 percent filling by November 1, 2022, and 90 percent filling by subsequent winters.¹⁸⁰ In addition, gas storage infrastructure is now considered critical infrastructure and is under scrutiny from EU authorities to avoid external interferences. The new gas storage regulation establishes coordinated gas refueling operations, including joint procurement, aggregation of orders and supplies, and optimization and improved transparency of gas infrastructure. REPowerEU proposed coordinated demand reduction strategies and a solidarity mechanism in the European Union in case of gas supply disruptions.¹⁸¹ It was accompanied by a recommendation for member states to take voluntary measures to maintain a collective reduction in gas demand of 15 percent, compared to average demand between April 2017 and March 2022.¹⁸²

On the natural gas side, REPowerEU established a framework for collective procurement of non-Russian LNG, extending participation to all interested member states, as well as Georgia, Moldova, Ukraine, and the Western Balkans.¹⁸³ The idea behind this platform, which is already functioning, is to increase transparency in the EU gas market, complementing the existing hubs to bring as many gas consumers as possible closer to the wholesale market and attract new operators to the growing European LNG market. In particular, the platform hopes to increase the bargaining power of landlocked European countries, such as Austria, the Czech Republic, and Slovakia, in their process of energy diversification and access to import terminals in other member states. The European Union's joint gas purchasing platform, AggregateEU, has shown some initial success in matching gas supply with demand, attracting significant participation from companies and suppliers in its first rounds.¹⁸⁴

However, its overall effectiveness remains uncertain due to limited transparency on the contractual outcomes and prices achieved through the platform. While the initiative has facilitated connections between buyers and sellers, the added value over existing market mechanisms is questionable, as the current EU gas market already supports effective demand aggregation and access to LNG supplies for landlocked countries thanks to EU rules on third-party access to pipelines.¹⁸⁵ Despite this, AggregateEU, originally designed as a temporary mechanism, is now a permanent facility incorporated into the EU gas and hydrogen package. Similar schemes are expected in the future to aggregate purchases of hydrogen and critical minerals at the EU level.¹⁸⁶

Transatlantic Cooperation

As the magnitude of the security and energy crisis Europe faced dawned on leaders on both sides of the Atlantic, they quickly initiated effective transatlantic cooperation on energy matters. Yet the energy dimension of the crisis meant that the United States needed to engage with the European Union as opposed to NATO. Relations between the United States and EU institutions have occasionally been fraught, and the United States has rarely prioritized them. Yet the crisis changed the dynamic, with the United States and European Union engaging extensively, especially on energy and sanctions.

A March 2022 joint U.S.-EU announcement established a new Task Force on European Energy Security.¹⁸⁷ This grouping has since met 11 times and has been an effective vehicle for transatlantic communication and coordination on energy issues, enabled by high-level leadership and participation on both sides. Task force members include Björn Seibert, head of cabinet for the European Commission president; Amos Hochstein, U.S. senior adviser for energy security; Ditte Juul Jørgensen, European Commission director general for energy; and Sarah Ladislaw, special assistant to the president and senior director for climate and energy at the White House National Security Council.

The task force has enabled much more robust U.S.-EU engagement in relation to the energy dimension of this crisis. U.S. LNG has been a major topic of the task force, including discussing "global LNG markets and market projections, the regulatory environment and permitting outlook in the United States and the EU, the development of U.S. LNG export capacities, the reinforcement of EU LNG infrastructure, and the EU Energy Platform and Joint Purchasing."¹⁸⁸

EU Sanctions on the Russian Energy Sector

The European Union has phased in sanctions on the Russian energy sector (Figure 17), seeking to minimize the negative impact on EU member states while still affecting Russian finances. EU law mandates that member states in the Council of the European Union unanimously approve sanctions, requiring the acquiescence of all EU states in order to come into force. The mandate has substantially limited the ambition of this instrument, holding back sanctions on Russian gas or nuclear fuel and requiring several country-specific exceptions when sanctioning crude oil or oil products. Sanctions have sought to reconcile national interests within the European Union and have run parallel to the unfolding war and the energy crisis.



Figure 17: EU Sanctions on Russian Imports Evolution, 2022-24

Source: Authors' analysis.

The first round of sanctions focused on limiting the access of Russian banks to the international financial system and the transfer of technology to the Russian energy sector. Coal was the first Russian energy source that the European Union sanctioned. The ban on imports of all Russian coal was agreed to in April 2022 as part of the fifth package of sanctions against Russia, granting a four-month grace period for existing contracts. Coal sanctions were relatively easy to agree upon thanks to a high level of self-sufficiency in Europe (75 percent in 2021), the short-term nature of coal contracts (around 90 days), and the relative availability of alternative suppliers in the Atlantic basin (e.g., Colombia, South Africa, and the United States).¹⁸⁹

The case for EU sanctions on Russian oil was more complicated. Sanctions on seaborne oil imports were agreed to in June 2022, with a six-month grace period and an opt-out provided for Bulgaria,

exemplifying the complexities of negotiations in the Council of the European Union. At the time of the negotiations, Russian state-owned Lukoil's refinery in Burgas, Bulgaria, was the largest industrial complex in the country, originally designed to process exclusively Ural-grade crude oil. Following sustained negotiations, Bulgaria obtained a "'special derogation' from the measure until the end of 2024 due to its 'geographical exposure'" as a means to achieve unanimity in the council.¹⁹⁰ The exception subsequently proved a loophole for sanctions evasion, and Bulgaria voluntarily withdrew from the exemption.¹⁹¹ Bulgaria is now seeking options to sell the Lukoil refinery to new owners willing to inject the investment necessary to allow the facility to process non-Russian crude varieties.

Russian oil pipeline imports have also been temporarily exempted from EU sanctions. Central European countries, especially the Czech Republic, Hungary, and Slovakia, citing their landlocked geography and the difficulties of switching their refineries to different crudes, obtained a temporary allowance for imports from the Druzhba pipeline. These countries, which continue to import Russian crude oil through this infrastructure, state that they will continue to do so in the medium term while building new import pipelines and upgrading their refining capacity.

Germany and Poland agreed to eliminate their temporary exception and cease imports of Russian crude oil by pipeline on January 1, 2023. In exchange, the commission allowed Kazakhstan to export oil to Germany through Russian pipelines. Although this technically is a move to eliminate Russian crude imports, Kazakhstan's oil, which must be supplied through the Russian transmission system, inevitably mixes with Russian oil, and Moscow receives oil transit fees.¹⁹² The temporary measure allows inland German refineries to operate at full capacity and avoid bottlenecks in the pipeline transportation system from the seaports.

An embargo on Russian-refined products (e.g., gasoline, diesel, and kerosene) came into force in February 2023 and curbed EU imports of most Russian petroleum products. However, non-EU countries can import Russian oil, refine products, and export these to the European Union, in effect creating a loophole for indirect imports of Russian oil. The Centre for Research on Energy and Clean Air has called China, India, Singapore, Turkey, and the United Arab Emirates "laundromats" for Russian oil, importing larger quantities of crude oil since the EU embargo while also growing export volumes of diesel and jet fuel to the European Union.¹⁹³

In addition to banning seaborne crude oil imports, the European Union has imposed several prohibitions regarding maritime services and access to its ports in relation to the oil trade between Russia and third countries. EU operators are barred from providing maritime services and insurance unless the trade adheres to the Group of Seven (G7) oil price cap.¹⁹⁴ Vessels engaging in ship-to-ship transfers suspected of violating the Russian oil import ban or the G7 price cap are denied access to EU ports, as are vessels that manipulate or deactivate their navigation tracking systems.

These restrictions have not been applied to LNG shipping services and transport. European terminals are offering transshipment services to Russian gas tankers. From November to June, during the Arctic winter season, most Russian LNG coming from the Yamal region needs icebreakers to access international markets. To optimize the routes of these specialized vessels, the natural gas

they transport is transferred to conventional LNG vessels, principally at the terminals of Zeebrugge in Belgium and Montoir-de-Bretagne in France, for subsequent export to non-European markets.¹⁹⁵ The eleventh round of European sanctions on Russia included transshipments of petroleum and derivatives that did not comply with the maximum price stipulated by the G7, but it deliberately omitted transshipment of LNG.

Package	Date	Coverage				
Second	February 25, 2022	Financial sector sanctions on 70 percent of the Russian banking system; prohibition of the sale, supply, transfer, or export to Russia of specific goods and technologies in oil refining.				
Third	March 2, 2022	Key Russian banks excluded from the SWIFT system in coordination with the United Kingdom and United States				
Fourth	March 15, 2022	Import ban on iron and steel products; prohibition of new investments in the Russian energy sector by EU firms				
Fifth	April 8, 2022	Import ban on all forms of Russian coal (four-month phase-in period); export bans on technology for liquefaction of natural gas.				
Sixth	June 3, 2022	Ban on seaborn crude oil imports (six-month phase-in period) and oil products (nine-month phase-in); exception for pipeline oil imports.				
Eighth	October 6, 2022	Implementation of the G7 oil price cap: European operators allowed to carry out and support transport of Russian oil to non-EU countries, provided its price remains under a preset cap				
Tenth	February 25, 2023	Import bans on bitumen, asphalt, synthetic rubber, and carbon black; prohibition on Russian nationals and entities booking gas storage capacity in the European Union (LNG excluded); measures to facilitate EU energy company divestment from Russia				
Eleventh	June 23, 2023	Import ban of Russian oil by pipeline for Germany and Poland; allowance of functioning of Caspian Pipeline Consortium pipeline, which transports Kazakh oil to the European Union through Russia; restrictions on port access to oil tankers transporting Russian oil				
Twelfth	December 18, 2023	No Russia clause (bans reexportation to Russia and for use in Russia of certain goods); prohibition on import of liquefied propane; Russian imports of pig iron and ferroalloys banned				

Table 1: Key E	U Energy	Sanctions	on Russia	since	February	2022
	01					

Source: Authors' analysis.

EXPLAINING THE LACK OF NUCLEAR SANCTIONS

In contrast to the drop in oil and gas trade between the European Union and Russia, imports of nuclear fuel products have steadily increased since February 2022. The absence of sanctions may be explained by the European Union's unique dependence on Russian nuclear fuel products

and, secondly, by the limited impact that sanctions have had on Russia's trade balance and the Kremlin's budget.¹⁹⁶ According to Eurostat, the European Union imported only €572 million of Russian nuclear-related products in 2022. The European Union has continued purchasing nuclear technology and fuel from Russia for the 19 Soviet-designed water-water energetic reactors (VVERs) operating in five member states: Bulgaria, the Czech Republic, Finland, Hungary, and Slovakia. Replacing Russian fuel is not a simple task: it is not an easily substituted commodity, and it requires adapting capacities in nuclear power plants and obtaining alternative suppliers in a very tight market. EU countries have increased Russian nuclear fuel deliveries from 314 tons of nuclear fuel in 2022 to 573 tons in 2023, a result that may be due to stockpiling in case of expanded conflict or supply disruptions.

Hungary is the only member state without explicit plans to transition to alternative fuel suppliers. The rest are pursuing alternate suppliers.¹⁹⁷ The Czech Republic is poised to begin receiving Western-manufactured nuclear fuel from Westinghouse and, potentially, Framatome starting in late 2024. Slovakia, through its nuclear plant operator Slovenské Elektrárne, aims to introduce Westinghouse fuel by 2027 following a licensing agreement, although a newly elected Russiafriendly government may negatively affect the implementation of these plans. Finland is preparing to shift its Loviisa reactor units to non-Russian fuel between 2027 and 2030, having engaged in discussions with Westinghouse about fuel supply even before the onset of the war.

EXPLAINING THE LACK OF GAS SANCTIONS

As of May 2024, the European Union has not imposed specific sanctions on Russian natural gas. While pipeline gas imports have fallen by four-fifths from 2019 levels following Russia's weaponization of gas supplies, LNG imports from Russia remain above prewar levels.¹⁹⁸ Sanctions on Russian gas have been a politically sensitive issue in Brussels. The unanimity required to impose sanctions is difficult to achieve due to the threat of further disruption of supplies disproportionately affecting countries most dependent on pipeline imports. Hungary's opposition is an example of a significant obstacle to agreement in the Council of the European Union. Hungarian prime minister Viktor Orbán has sought to engage and work with the Kremlin even after the invasion, reaching an agreement for the purchase of additional volumes of gas and the diversion of its Russian gas imports away from Ukrainian pipelines to the TurkStream pipeline.¹⁹⁹

In its recent Regulation on Gas and Hydrogen Networks, the European Union has implemented a mechanism which allows member states to temporarily block Russian and Belarusian exporters from reserving infrastructural capacity for natural gas shipments via pipeline and LNG.²⁰⁰ This mechanism, which has to be implemented voluntarily at the state level, would allow European companies to terminate their contracts by invoking force majeure clauses and protect themselves against possible arbitration. Because this mechanism does not constitute an EU-wide ban or sanction, it circumvents the unanimity standard. While European commissioner for energy Kadri Simson has urged member states to use this mechanism, it is uncertain whether the mechanism will be implemented in a coordinated and effective way.

While Europe actively sought to decouple from Russian coal and oil, it did not collectively seek to decouple from Russian gas. Instead, Europe has been forced to scramble due to the collapse in supply. Europe has not sanctioned Russian gas, though it is now considering sanctioning LNG. The reduction in purchases of Russian natural gas has been mainly due to lower volumes shipped through Nord Stream (out of operation) and Yamal (closed due to the Kremlin's requirement to make payments in rubles). The transit contract through Ukraine for Russian gas expires on December 31, 2024.²⁰¹ Its future is unknown. Although Ukraine has announced its refusal to negotiate bilaterally with Gazprom, it has opened the door to possible contracted volumes on the spot market for interested European countries—mainly Austria, Hungary, and Slovakia. This uncertainty about the future of Ukrainian pipelines and the creation of a possible European mechanism to block Russian LNG imports should help to further reduce Russian gas imports in 2025 and 2026.²⁰²

Postcrisis Policy Landscape

Toward a New European Energy Security Strategy?

I urope came through the crisis, but challenges remain. From a supply perspective, Europe has found a way to replace Russian gas. But these are largely from sources external to the European Union, and the costs of these alternative sources are higher and will therefore impact Europe's economic competitiveness.²⁰³

Electricity Market Reform

The electricity sector emerged from the 2022 energy crisis as a major locus of energy security considerations for policymakers. The surges in electricity pricing devastated consumers, both households and businesses, and cascaded shocks from the natural gas sector throughout the economy.

The status of marginal pricing sits at the center of the debate. During the crisis, numerous ad hoc interventions into the electricity market were proposed; some were implemented to reduce consumer exposure to spot prices set by marginal pricing. After the crisis, the sector emerged with newfound political relevance, which catalyzed widespread interest in reforming markets based on the lessons of 2022. Nonetheless, widespread consensus on the need for review and reform in the sector does not imply widespread agreement on the substance of that reform.

Two key objectives appear in conflict. On one hand, reform seeks to limit the exposure of electricity consumers to volatile pricing sourced from gas-fired generation and their ability to benefit from the low prices of new renewable generation. This implies limiting consumer exposure to prices set via marginal pricing in the market. On the other hand, massive volumes of investment must be

attracted to the sector to make continued progress on decarbonization. This includes investment in wind and solar generation, but expanded investment is of increasing importance in technologies such as battery storage and hydrogen-fired generation, which can help balance the intermittency of renewables. Broadly stated, price signals, free of regulatory risk and reflective of overall system scarcity, are the best way to attract private capital at scale. Furthermore, price signals are the crucial incentive for demand response programs, the expansion of which is a crucial objective for overall sectoral efficiency and managing the intermittency of renewable energy.



Elena Giannakopoulou, group chief strategy officer of Public Power Corporation S.A., Greece, speaking at the 2024 CSIS Energy Security and Geopolitics Conference.

"We are moving from a fuel-based system to a technology based one, but this change is not happening overnight in the sense that you need to manage the risks and the complexities.... And as we are planning our transition to a technology-based system where materials, equipment, manufacturing capacity, critical raw materials are key in this sense, we ... are trying both in terms of technologies and in terms of geographies (to find) complementarity that we can leverage if we operate in different systems. And of course, this has a big impact on the energy security of a ... region like Europe."

> — Elena Giannakopoulou, Group Chief Strategy Officer, Public Power Corporation S.A., Greece

An initial package of reforms was proposed in March 2023, and following a sustained period of negotiations, the Council of the European Union approved a final package in March 2024.²⁰⁴ Overall, this package yielded considerable incremental reform while resisting more radical changes that sought to move away from a single marginal market clearing price determined by merit order dispatch.

The package empowers EU member states to deploy fixed-price and fixed-term contracts, which can come in the form of power purchase agreements or contracts for differences (CFDs).²⁰⁵ Both contracts are already in use across EU member states to varying extents, but the new rules direct member states to make these contract arrangements available to consumers and project developers and, in particular, allow for contracts that cross national borders.

The rule also creates room for member states to deploy new capacity support mechanisms. In U.S. electricity markets such as PJM, capacity markets are a well-established mechanism to manage long-term resource adequacy. They create a second vector, independent of spot marginal pricing, by which funds can flow from consumers to generators organized around the principle of availability at periods of peak demand. Germany has proposed exactly such a mechanism to attract investment into the hydrogen-fired thermal power plants it needs to build to balance out renewables.²⁰⁶

Uncertainty remains over how member states will interpret and implement the EU reform package. The proposal invokes various design principles, with which both capacity mechanisms and CFDs should comply. The exact technical limits of these principles and the enforcement of these limits remain to be seen and will determine the overall impact of the reform package. Overall, the package seeks a tenuous balance between further integration of a competitive European electricity market to the benefit of efficient decarbonization and the need to preserve member states' ability to procure generation resources in line with their specific national strategies.

Banking on Hydrogen: The EU Hydrogen Strategy and Diplomacy

Key to Europe's energy security and climate ambitions are the development and deployment of green hydrogen. Hydrogen was already an essential component of the EU decarbonization strategy before Russia invaded Ukraine. However, it took on a renewed and more prominent role in EU energy plans as the bloc sought to diversify away from Russian fossil fuel resources after the invasion.

As mentioned in previous chapters, REPowerEU was conceived at the height of the energy crisis to diversify the European Union away from Russian fossil fuels.²⁰⁷ By updating the Fit for 55 targets published only a year before, the initiative further galvanized the bloc's clean energy ambitions. For hydrogen, it brought about two landmark changes. First, the European Union began targeting a market of 20 million tons of renewable hydrogen by 2030, doubling its initial commitment of 10 million tons produced by member states by importing 10 million tons from reliable suppliers across the globe. Second, support for low-carbon hydrogen derived from natural gas was tacitly phased out, and plans for government funding were scratched. This lack of support was a ramification of natural gas supply disruptions during the energy crisis jeopardizing hydrogen-dependent industries, such as ammonia producers and refining operations, as well as the premise that high gas prices made low-carbon hydrogen obsolete as a transitional stand-in for renewable hydrogen.²⁰⁸

To match the doubling of supply targets, demand targets for end-use sectors received a commensurate boost (Figure 18). Industry saw an increase in the share of renewable hydrogen consumption in the sector from 35 percent to 75 percent by 2030. Meanwhile, the transport

sector nearly doubled its 2030 target for consumption of renewable fuels of nonbiological origin. The Hydrogen Valleys initiative also had its funds topped up in hopes of doubling the number of hydrogen valleys in the European Union by 2025.



Figure 18: Hydrogen Consumption 2030 Targets by End-Use Sector

Source: "Commission Staff Working Document: Implementing the REPowerEU Action Plan: Investment Needs, Hydrogen Accelerator and Achieving the Bio-Methane Targets," European Commission, May 2022, https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=SWD%3A2022%3A230%3AFIN.

However, these objectives have faced substantial criticism due to the unrealistic targets on both the supply and demand sides. A recent report warns that the European Union is not on track to deploy enough renewables or electrolyzers to meet its production targets by 2030, while the outlook on imports is still uncertain.²⁰⁹ Meanwhile, consumption targets rely on unconsolidated end uses and lagging adoption in existing demand sectors.²¹⁰ Industry executives have pointed out that the high cost of renewable hydrogen proves a major barrier to wider adoption, warranting a reassessment of initial targets.²¹¹

SECURING SUPPLIES THROUGH HYDROGEN DIPLOMACY

Already a core component in the EU hydrogen strategy, the international dimension has become central to achieving REPowerEU's goal of importing up to 10 million tons of renewable hydrogen. Through so-called "green hydrogen partnerships," the European Union will not only seek to meet its import quota while supporting clean energy project development in partner countries but also unlock the generational opportunity to rewire its energy partnerships and diversify its resource dependencies. REPowerEU envisions the development of three hydrogen corridors via the Mediterranean Sea, the North Sea, and through Ukraine, once conditions allow.²¹² This ambition has led the European Union through the development of strategic partnerships with Azerbaijan, Egypt, Kazakhstan, Morocco, Namibia, Norway, and Oman, among others.²¹³ The European Commission has also held talks with Saudi Arabia to increase energy cooperation, with hydrogen as a central pillar.²¹⁴ On the transatlantic front, the European Union has already developed a hydrogen partnership with Canada and is looking to expand those partnerships into Latin America.²¹⁵ Lastly, the United States is poised to become one of the leading renewable hydrogen markets, thanks to the incentives and tax credits contained within the Inflation Reduction Act (IRA). In a bid to share this cornucopia, the Mission Possible Partnership is looking to spur U.S.-EU hydrogen trade opportunities by enabling Europe's first shipment of clean hydrogen from the United States in 2026.²¹⁶ All of these partnerships offer the European Union plenty of avenues for renewable hydrogen import supply diversification.

New Dependencies? Risks Posed by Clean Tech Supply Chains

As Europe seeks to further accelerate a clean transition that improves its overall energy security position, it faces new dilemmas in the domains of trade and economic security. The rapid expansion in renewable energy development has been characterized as merely shifting dependencies from Russia to China. As described by U.S. assistant secretary for energy resources Geoffrey R. Pyatt at the 2024 CSIS Energy Security and Geopolitics Conference, "We need to make sure that we don't replace an era of dependence on Russian fossil fuels with dependence on critical minerals and critical mineral processing in the People's Republic of China."²¹⁷

The new dependencies are real. Europe is 95 percent import dependent on China for solar PV panels.²¹⁸ The European wind sector, which has for a long time been self-supplied, is coming under intense competitive pressure from Chinese firms and could be competed out of existence in the way German solar manufacturing was by Chinese firms.²¹⁹

These actual and potential import dependencies pose challenges for policymakers. However, analytical clarity demands they be classified as trade or economic risks rather than energy security risks. Unlike gas or other fuel imports, there is very little risk to the short-term stability of the European energy system from a hypothetical interruption in access to Chinese solar panels. In short, Europe is currently trading reduced energy security risk for increased trade and economic risks.

The scale of trade and the economic risks are complicated. The availability of ultra-low-cost (and increasingly cheaper) Chinese solar panels and wind turbines lowers the overall investment cost of Europe's climate ambitions. The value here should not be overlooked given the massive fiscal outlays the energy crisis imposes; for the European Union, the efficient use of public funds and subsidy capacity are ever more important.

Inversely, allowing Chinese imports to overwhelm and destroy the viability of European clean manufacturing companies has economic, social, and political costs. Expanding the manufacturing employment base associated with climate technologies to improve the overall standing of climate

politics is the central political-economic thesis of the IRA and is a model with considerable support in Europe (especially in the European Commission).²²⁰ Allowing Chinese imports to displace extant industries and jobs is likely to cause political backlash that undermines EU climate ambitions.



Ben McWilliams, an affiliate fellow with Bruegel, speaking at the 2024 CSIS Energy Security and Geopolitics Conference.

"Trying to define an industrial strategy in Europe of what we want to produce and how we want to produce it in this global competition between America and China ... is a really hard thing to do [when you need] unanimous agreement among 27 EU member states. And in my mind, that's the kind of shaping energy and climate question of the day."

- Ben McWilliams, Affiliate Fellow, Bruegel²²¹

Trade and economic competition concerns extend beyond China. The IRA implemented subsidies for clean energy manufacturing and successfully attracted large volumes of private capital investment. These subsidies apply to countries that hold a free trade agreement with the United States (e.g., Canada and Mexico). Notably, the European Union lacks any such free trade agreements. Thus, European-based manufacturing cannot participate in these subsidies–most notably, the electric vehicle tax credit.

The trade and economic risks of clean energy supply chains extend past technology and into the domain of raw materials. Europe is structurally short of the minerals critical to most modern clean technology. For example, the European Union is solely dependent on one country for heavy rare-earth elements (100 percent supplied by China), boron (98 percent supplied by Turkey), and

platinum (71 percent supplied by South Africa). Recognizing that the bloc will never be self-sufficient but should diversify its supply of critical raw materials, the European Council adopted the European Critical Raw Materials Act in March 2024.²²² Again, the direct comparison to fuel dependencies is inappropriate and represents a trade and economic risk rather than an energy security risk. Although Europe has some domestic mining and processing capacity–notably in the Nordic states– it is heavily dependent on raw materials and, more extensively, on processed materials used as inputs in higher-value manufacturing.

Another vector of risk arrives through expanded Chinese clean energy investments in Europe. On a recent trip to Europe, Chinese president Xi Jinping traveled to Hungary, where it was recently announced that China would be opening new battery factories.²²³ China appears to be cultivating Hungary as a bastion of support in the European Union in the hopes that it will shield China from potential joint EU measures, akin to the role Hungary has played vis-à-vis Russian sanctions. Hungary has frequently blocked EU statements condemning or opposing Chinese action. In this respect, although clean energy supply chains can help reduce European energy security risks, they invite considerable risks in the domain of trade policy and, perhaps more troubling, undermine Europe's ability to form coherent foreign policy.

New and Different Gas Risks

Overall, the Russian gas "weapon" is unlikely to ever pose the energy security risk to Europe that it posed before the crisis. This holds true even though Russian gas continues to flow into the European market, albeit at greatly reduced volumes, through both pipelines and LNG imports. Commercial developments, infrastructure investments, and political commitments have structurally reduced European reliance on Russian gas, and volumes are unlikely to ever return to precrisis levels.

While many businesses may be open to a return to Russian gas, political endorsement or support for this strategy by European policymakers today appears unlikely so long as Putin remains in the Kremlin. For Europe to reengage Russia as an energy supplier, major political change in Moscow or a favorable end to the war that preserves Ukraine's path to Europe is likely necessary.

Diversification is the most reliable principle by which to measure energy security, as established most famously by British statesman Winston Churchill in 1920: "Safety and certainty in oil lie in variety and variety alone." Today, the diversification of suppliers, import vectors, and regional exposure of natural gas imports has to some extent structurally improved European energy security. However, it would be naive and simplistic to think Europe is free of gas-derived energy security risks. Today, European exposure has shifted elsewhere. Norway's leading role in European gas markets makes sabotage against Norwegian undersea infrastructure a risk; such an event, timed with winter or storage-filling season, could drive gas shortages and price spikes in Europe.

Additionally, Europe continues to rely on imports from states in North Africa and the Middle East, which could easily present diplomatic complications and provide leverage against European objectives. While recent Houthi attacks in the Red Sea have demonstrated the vulnerabilities of sea lanes and LNG supply from the Persian Gulf, pipeline supply remains the most complex to manage.²²⁴ Gas from Algeria and Libya, accounting for more than 10 percent of EU imports, faces problems of flat or declining production, rapid growth in domestic consumption, and geopolitical risks. At least since the Arab Spring, the European Union has demonstrated profound limitations in influencing the Maghreb and building an attractive neighborhood narrative. In the case of the Southern Gas Corridor, Azerbaijan's bellicose attitude in its conflict with Armenia presents a risk to EU interests and the defense of its principles in the Caucasus.²²⁵

Finally, Europe now relies heavily on energy imports from the United States. Although the recent approval pause of a U.S. LNG export terminal does very little to affect current LNG flows and may very well do little to affect future flows, the development highlights how LNG exports have become a pawn in domestic politics. A hypothetical future with a climate-ambitious president in the White House with control of Congress could easily see scaled-back U.S. LNG exports, a result European climate advocates likely will applaud even as it may hurt European energy security. The nature of shale gas, based on short-cycle projects, means that without constant upstream investment, production may plateau or decline, jeopardizing the viability of continued expansion of LNG exports.

Today, Europe is structurally exposed to global LNG markets in a way it never was before the crisis. Developments anywhere in the world that affect global LNG markets will affect the price and availability of LNG for Europe. This new exposure demands new, sustained, and serious attention from European policymakers.

Building New Connective Midstream and Integrating Europe's Energy Market

The energy crisis has demonstrated the considerable reliability and efficiency value of integrated infrastructure and market networks. It also demonstrates how much further integration is possible and needed. The combined pressure of an ongoing energy decoupling from Russia and an accelerating energy transition requires Europe to invest in physical infrastructure–both electric transmission and pipeline networks–and advance the integrated market systems that extract maximum value out of these networks.

TRANSMISSION

Electric transmission played a crucial role in enabling European member states to avoid loadshedding situations through the summer of 2022 and the winter of 2022-23. Variable renewable energy (e.g., wind and solar) allows for reduced reliance on Russian gas in the power sector but requires greatly expanded intra- and intermember state transmission infrastructure to reduce curtailment rates and price cannibalization effects and allow for continued rapid deployment.



Amy Myers Jaffe, a research professor at New York University, speaking at the 2024 CSIS Energy Security and Geopolitics Conference.

"We found ourselves in is this situation where suddenly NATO is having to police interconnector wires between different countries in Europe.... Managing the geopolitical complications of making a system where everybody has to have a certain amount of backup generation—and we're going to utilize it strategically as an entity—[entails] data and physical challenges."

— Amy Myers Jaffe, Research Professor, New York University²²⁶

The 2022 REPowerEU plan targets expanded electricity interconnection projects that cross member state boundaries and improve European electric integration. In November 2023, the European Commission proposed an action plan aimed at accelerating transmission project development and announced a list of 166 Projects of Common Interest (PCIs), 88 of which involved the electric system.²²⁷ The PCI designation grants projects considerable permitting advantages and makes them eligible for EU funding from the Connecting Europe Facility (CEF).²²⁸ Notable examples of projects underway include the France-Spain interconnection and the Baltic synchronization project.²²⁹

Transmission projects, especially large, high-voltage interstate projects that deliver the most system value, are very costly and difficult to plan, permit, and build. Cross-border projects require difficult political coordination among member states and the ever-present challenge of social acceptability.²³⁰ Although planning and policy are improving and overall investment in grids is growing, research by Ember indicates that the grid investment plans of many European utilities fail to match member state renewable energy policy targets.²³¹ Cost is the primary brake on expanded investment for both utilities and member states because the investment dollars are paid by rate payers who are already reeling from greatly elevated electricity prices. The CEF is funded at €5.84

billion through 2027, but the commission estimates €584 billion in investments are needed in the electric grid by 2030.²³² The French-Spain interconnection project costs €2.8 billion alone, and Europe needs dozens of such projects if targets are to be met.²³³ In short, expanded funding at the EU level is likely necessary to achieve the European Commission interconnection target of 15 percent by 2030.²³⁴

PIPELINES

On the fuels side, the considerable investment and expansion needed in pipeline networks pose a unique challenge. Many member states continue to rely on Russian piped oil and gas due to infrastructure constraints. Investment in new and expanded pipeline networks will help member states decouple further from Russia and improve overall European energy security and market stability. At the same time, if EU climate ambitions are to be realized, Europe must transition away from these fossil fuel networks. Right-sizing investment to achieve near- and medium-term energy security objectives without jeopardizing long-term climate targets is a difficult needle to thread.

The energy crisis has compelled the European Union to quickly invest in new gas infrastructure– both import terminals and internal transport capacity. However, these investments come with risk of carbon lock-in, as existing infrastructure tends to be used, thus delaying European energy transition objectives.

The generally agreed-upon solution is in dual-use infrastructure: today's gas infrastructure should be built such that it can be used for the hydrogen networks of tomorrow. Although reasonable in concept, the engineering in practice is more complicated. LNG terminals in particular face challenges related to adaptation to low-carbon gases, such as hydrogen or low-carbon ammonia. The technical and economic feasibility of such conversions is still at an exploratory stage, with significant uncertainties surrounding the costs and efficiency of the conversion processes.²³⁵

In delivering a substitute for Russian fossil-gas, the European Union's strategic bet on decarbonized hydrogen has clear geopolitical and energy security benefits alongside obvious climate benefits. But the development of a functional hydrogen market and industrial ecosystem requires dedicated infrastructure to connect production and consumption centers, similar to the expansive existing gas transportation network. EU transmission system operators have proposed a hydrogen backbone to repurpose existing infrastructure dedicated to natural gas and complement it with new hydrogen-specific pipelines and storage sites.²³⁶ A major challenge is the unknown scale, composition, and geographic distribution of future hydrogen production and consumption centers. Investment of public or private dollars is therefore difficult to justify. Some mega projects, such as H2med, connecting Barcelona to Marseille via an undersea pipeline, have already been criticized for their lack of economic realism, sheer technical difficulty, and weak climate credentials.²³⁷ The sum of technology and market uncertainty and member state policy coordination challenges means that development of a European hydrogen infrastructure network and market will likely take longer and be more costly than assumed.

Protecting Critical Energy Infrastructure

The 2022 Nord Stream sabotage revealed the vulnerability of Europe's critical energy infrastructure in the undersea domain. In October 2023, a group of undersea pipelines and telecom cables linking Finland to Estonia were also damaged and taken out of operation.²³⁸ In this case, the cause was identified as a Chinese ship dragging an anchor, although explicit motives have not been established.²³⁹ Europe is structurally exposed to hybrid threats such as these against the extensive undersea infrastructure networks in the Baltic and North Seas. Oil and gas pipelines are the traditional class of infrastructure in these waters. With the increased importance of Norwegian gas flows, the energy security imperative to protect these assets has grown.



Emily Holland, assistant professor at the U.S. Naval War College, speaking at the 2024 CSIS Energy Security and Geopolitics Conference.

"One of the big lessons that came out of the energy crisis, and that still is coming, is that we did not have enough resilience in terms of our energy infrastructure... . Things could very easily come down. And that's not energy security.... So, we need to build overcapacity. And to do that, we need to convince the private sector and insurers to keep building these assets."

- Emily Holland, Assistant Professor, U.S. Naval War College²⁴⁰

Although Russian ground forces have been depleted in Ukraine and Russia's military industrial base has been stretched, the country's capacity to inflict damage on European energy infrastructure remains intact. While Russia's navy has been underfunded for years and consists largely of Sovietera platforms, its underwater capacity continues to grow. Russia's submarine program has been a noteworthy priority, exemplified by the Kremlin authorizing 13 new nuclear and conventional submarines since 2014. A CSIS report from 2023 notes, "Russia's ability to target critical infrastructure short of war and impose economic costs to deter external intervention in regional conflicts is an important component to Moscow's doctrine and thinking on escalation management."²⁴¹

Meanwhile, the energy transition implies massive investment in offshore wind and undersea transmission in the Baltic, Mediterranean, and North Seas. Attacks on undersea electric transmission cables have not been observed to date but could create power grid disruptions that cascade failure across the network. Thus, the energy transition toward clean energy assets in many ways expands undersea critical infrastructure exposure. NATO has established a new operational group, the Maritime Centre for Security of Critical Undersea Infrastructure, to manage these risks. Appropriate levels of risk management likely require increased funding, as Europe and NATO will need to procure more operational assets to expand coverage, including traditional surface fleet assets alongside new technologies such as sea drones, sensor networks, and the use of AI.²⁴²

Expanding Transatlantic Energy Cooperation

The energy crisis has demonstrated the value of sustained transatlantic cooperation and coordination. During the crisis, European access to U.S. LNG was critical. After the crisis, Europe will remain reliant on U.S. LNG volumes over the medium term as it seeks to cement its energy decoupling from Russia. The U.S. LNG permitting pause has further demonstrated the value of established and close working relationships; policy decisions and impacts on both sides of the Atlantic can be accurately communicated and discussed, and diplomatic rifts can be avoided.

The value of reliable partnerships in energy trade highlights the need and potential for greater U.S.-EU cooperation when it comes to broader clean technology and climate policy issues. Despite considerable U.S.-EU engagement in this area, there have been few breakthroughs. Sustained efforts to negotiate a green steel agreement have been unsuccessful.

The passage of the IRA in the United States caused considerable diplomatic backlash in Europe, as it was felt that the policy unduly discriminated against European suppliers. The Biden administration has sought to soften the impact of the IRA on European suppliers, but there has been no movement on either side of the Atlantic toward a new U.S.-EU trade agreement despite the extensive engagement established through the U.S.-EU Trade and Technology Council.²⁴³

There are major energy and climate issues that require transatlantic partnership. The implementation of the European Union's Carbon Border Adjustment Mechanism will put in place tariffs on carbon-intensive imports into the European Union. The European Union and United States should consider this an opportunity to coordinate on carbon-linked trade. They would also benefit from close coordination on climate and trade policies focused on China, such as with regard to electric vehicles, batteries, or critical minerals.

Lessons Learned

he crisis spawned many lessons relevant for experts and policymakers in a variety of fields. Most of the lessons are positive. Europe has come out of the crisis more energy secure and has demonstrated the resilience and strength of its European project. However, the crisis left a significant fiscal mark on European economies and, through price shock and inflationary effects, gave rise to more antiestablishment parties, many of which seek to roll back aspects of the energy transition.

- Democracies and markets are resilient and durable and can survive shocks. Europe showed tremendous strength and resilience in response to the energy shock. The energy shock became a test not just of Europe's energy sector but also of Europe's union and the resilience of its democratic political system, market economy, and free and open societies. Europe's markets responded, creating incentives for action, and Europe's allies and partners stepped up. In the face of strategic coercion, Europe took drastic action, leveraged its wealth and fiscal capacity, and demonstrated the resilience of its market-based economies. That response depended critically on the creativity of its democratic political system. Instead, a Eurobarometer survey from 2023 showed that 58 percent of EU citizens believed that the bloc should accelerate the green transition in view of the energy crisis triggered by Russia's invasion of Ukraine.²⁴⁴
- The strength of the European Union and its ability to act should not be underestimated. Heading into the crisis, all major actors underestimated the strength and resilience of the European Union, including Russia, the United States, and Europeans

themselves. The bloc's fiscal firepower could be mounted at scale, with speed, and with great effect. European fiscal capacity was leveraged just as it was during the Covid-19 pandemic to protect European consumers, businesses, and the European way of life. As a result, the European Union is a much more prominent energy actor than before the crisis, demonstrating that not only is the union forged in crisis, but it enables Europe to respond to crises much more powerfully and effectively than if countries operated independently. In that sense, the European Union is a source of Europe's resilience.

- **Russia failed in its attempt to blackmail EU member states.** Ultimately, Russia lost the energy war with Europe. Russia now has significant stranded gas assets, leading to significant losses for Gazprom. While Russia has benefited from high energy prices and has found ways to keep its oil on the market, it failed to leverage its dramatic reduction in gas flows into Europe before the Nord Stream explosions. Europe progressively increased its aid to Ukraine and expanded its sanctions efforts against Russia, including placing an embargo on Russian oil, which has had a more significant impact than the oil price cap mechanism.
- U.S. LNG was critical to mitigating the crisis and can play an important role in U.S. foreign policy. The export of U.S. LNG filled the huge gap left by Russian natural gas supplies. In an era of great power competition in which energy can be weaponized, the ability of the United States to supply countries with LNG through market forces can reduce the ability of Russia or other autocratic states to use energy to coerce.
- The market shock served to supercharge renewables deployment, particularly solar energy. While the shift to LNG and alternative gas imports was central to maintaining energy security in the wake of the crisis, the expansion of renewables was equally remarkable and demonstrated that the rapid deployment and expansion of renewables is possible. However, there is also a need to simultaneously integrate and modernize grids, as well as expand storage to better cope with the intermittency of renewables.
- There is economic urgency for Europe to accelerate the green transition. The loss of access to lower-cost Russian piped natural gas in exchange for imported LNG has affected European economic competitiveness, with significant implications for Germany in particular. Thus, accelerating the energy transition for Europe is essential not just to meet climate goals or enhance Europe's energy security but also to ensure its economic competitiveness.
- U.S.-EU partnership is vital and must be strengthened. U.S.-EU engagement and coordination in these sectors have become much more robust and have laid the groundwork for a stronger U.S.-EU relationship. Historically, this relationship has been strained or limited. Different policy approaches to climate action, exemplified by the United States relying heavily on subsidies and the European Union pursuing a more balanced policy mix centered on carbon pricing, complicate the harmonization of transatlantic energy and climate policy. However, the importance of issues at the intersection of energy, climate, and trade will increase in the coming years, as the clean-tech revolution unfolds and becomes critical to the economic model of both blocs as well as to their wider geopolitical positioning vis-à-vis China. The U.S.-EU Trade and Technology Council is a promising forum for closer

coordination but has so far failed to resolve tensions stemming from each side's parochial interests on technology regulation and state aid rules. Finding policy alignment between the two blocs will thus continue to be a challenge but must be prioritized to achieve shared transatlantic objectives.

- The crisis has added to Europe's fiscal strains. Massive fiscal outlays were deployed to keep industry afloat, particularly in Germany. The inflationary impacts of higher energy costs had a deeply negative impact on European households and the European economy. The European fiscal response, which came after the massive fiscal outlays in response to the Covid-19 pandemic, has created a more difficult fiscal environment for European policymakers that presently limits their freedom of action. For instance, the dramatic policy ambitions of the government under UK prime minister Liz Truss were strongly repudiated by markets, forcing a policy retreat and her eventual resignation.
- **Political backlash came primarily against incumbents rather than Ukraine.** In 2022, there were fears that the shock of energy costs would turn Europeans against supporting Ukraine. That did not happen, and support for Ukraine remained high. Nevertheless, although European publics were aware of the cause of the energy crisis, political leaders have felt blowback. Discontent with economic conditions had negative repercussions on policymakers across Europe; even if there was only limited initial political blowback, the energy crisis was not politically cost-free.

The rise of far-right parties ahead of the 2024 European parliamentary election and the surge in farmer protests railing against environmental policies suggest that European leaders have limited latitude for ambitious climate action. However, public polling figures paint a very different picture. A 2023 Eurobarometer survey indicated that 86 percent of respondents found it important to take initiative at the European level to limit the immediate impact of rising energy prices on consumers and companies during the prior year.²⁴⁵ In terms of future action, 86 percent thought that gas storage within the bloc should be padded to avoid the risk of shortages, 85 percent believed that the manufacturing of clean technologies should be boosted, and 79 percent believed that EU measures should aim at reducing energy consumption.

Conclusion

he energy story that began with Russia's invasion of Ukraine has not ended. In the more than two years since the invasion, Europe has demonstrated its strength and resilience by emerging more energy secure, with the green transition relatively on course, and avoiding economic recession. But challenges still loom, as the rise in energy costs has created a systemic challenge for European industry and competitiveness. While European solidarity behind Ukraine has been maintained, and the threat of Russian energy extortion has been largely extinguished, the political fallout from the shock is still being felt. Anti-incumbent political sentiment remains high, leading to a surge in far-right parties. Opposition to Europe's climate agenda has grown and will likely result in a new European Parliament that is less supportive of strong climate action.

Thus, Europe faces cross-cutting pressures. On the one hand, there is a strong economic need to increase Europe's domestic energy supplies. For most European countries, aggressive deployment of renewables will be the most cost-effective course, balancing cost with climate ambition. But rising pressures could lead to backsliding. If high energy costs are blamed on the transition to low-carbon energy, they could inspire populist opposition to the "green agenda" and stall bolder action in the coming years. And the establishment of secure supply chains for clean energy technology remains a challenge, so that Europe does not swap one form of dependence for another.

About the Authors

Max Bergmann is the director of the Europe, Russia, and Eurasia Program and the Stuart Center in Euro-Atlantic and Northern European Studies at the Center for Strategic and International Studies (CSIS). Prior to joining CSIS he was a senior fellow at the Center for American Progress, where he focused on Europe, Russia, and U.S. security cooperation. From 2011 to 2017, he served in the U.S. Department of State in a number of different positions, including as a member of the secretary of state's policy planning staff, where he focused on political-military affairs and nonproliferation; special assistant to the undersecretary for arms control and international security; speechwriter to then secretary of state John Kerry; and senior adviser to the assistant secretary of state for political-military affairs. Before serving in the State Department, Bergmann worked at the Center for American Progress as a military and nonproliferation policy analyst and at the National Security Network as the deputy policy director. Bergmann holds a master's degree in comparative political science from Bates College. His recent commentary has been published in outlets such as *Politico*, the *Washington Post, Foreign Affairs*, and Sasakawa USA.

Cy McGeady is a fellow in the Energy Security and Climate Change Program at the Center for Strategic and International Studies (CSIS). He covers power markets, the energy transition, climate finance, and the macro trends affecting investment in the energy sector. Cy has years of experience working in the power sector across a series of roles at Genscape and Wood Mackenzie. Cy has experience consulting on project development, managing assets in short-term markets, and originating long-term transactions with a wide range of parties in the power sector. Cy has also closely studied transmission systems, generation dispatch, and price formation in power markets. Cy has recently worked in a capital advisory role with private funds seeking to invest in sustainability and climate solution projects. He is currently studying for a MA in liberal arts at St. John's College and holds a BA in economics from the University of Rochester.

Otto Svendsen is a research associate with the Europe, Russia, and Eurasia Program at the Center for Strategic and International Studies (CSIS), where he provides research and analysis on political, economic, and security developments in Europe. Prior to joining CSIS, Otto was affiliated with Albright Stonebridge Group, the Atlantic Council, and the National Democratic Institute. He holds an MSc in international relations from Georgetown University's School of Foreign Service and a BSc in international business from Copenhagen Business School.

Mathias Zacarias is a research associate in the Energy Security and Climate Change Program at the Center for Strategic and International Studies (CSIS). In this role, he supports research on hydrogen, power sector dynamics, clean energy supply chains, and methane abatement. Prior to joining CSIS, he worked as a graduate research assistant at the Georgia Institute of Technology, where he researched the potential of demand-side resources as a decarbonization pathway of the electricity

system, the shift in environmental impact priorities as the energy system continues to decarbonize, and the dynamics behind nascent building technology adoption. He also worked as an intern in the CSIS Energy Security and Climate Change Program, where he covered hydrogen deployment strategies. He holds an MS in civil engineering from the Georgia Institute of Technology and a BS in civil engineering with minors in sustainable cities and German from the Georgia Institute of Technology.

Ignacio Urbasos is a researcher in the Energy and Climate Programme at the Elcano Royal Institute in Madrid, Spain. He was previously part of the Energy and Climate Centre at the Institut Français de Relations Internationales (IFRI) where he developed his research into the investment strategies in the European electricity sector and energy policy in Latin America. He holds a bachelor's degree in international relations from the University of Navarra and a master's degree in international energy from Sciences Po Paris.
Endnotes

- 1 Fatih Birol and Ursula von der Leyen, "Europe Has Taken Its Energy Destiny Back into Its Own Hands," International Energy Agency (IEA), April 12, 2024, https://www.iea.org/commentaries/europe-has-takenits-energy-destiny-back-into-its-own-hands.
- 2 Ben McWilliams et al., "The European Union-Russia Energy Divorce: State of Play," Bruegel, February 22, 2024, https://www.bruegel.org/analysis/european-union-russia-energy-divorce-state-play.
- 3 Birol and von der Leyen, "Europe Has Taken Its Energy Destiny Back into Its Own Hands."
- 4 Max Bergmann et al., "Energy Security and Geopolitics Conference," (public event, CSIS, Washington, DC, April 23, 2024), https://www.csis.org/events/energy-security-and-geopolitics-conference.
- 5 Giovanni Sgaravatti et al., "National Fiscal Policy Responses to the Energy Crisis," Bruegel, June 26, 2023, https://www.bruegel.org/dataset/national-policies-shield-consumers-rising-energy-prices.
- 6 Bergmann et al., "Energy Security and Geopolitics Conference."
- 7 Ciaran Wark, "European Energy Prices Increase Up to 500% Compared to Pre-crisis Levels," GreenMatch, March 28, 2024, https://www.greenmatch.co.uk/blog/energy-prices-europe.
- 8 "The Schuman Plan Declaration," *European Community*, May 1970, 3, https://aei.pitt.edu/43776/1/A7519. pdf.
- 9 Joseph Dutton, "EU Energy Policy and the Third Package" (working paper, University of Exeter, UK, July 2015), https://ukerc.rl.ac.uk/UCAT/PUBLICATIONS/EU_energy_policy_and_the_third_package.pdf.
- 10 Commission of the European Communities, *The Internal Energy Market* (Brussels: Commission of the European Communities, May 1988), https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:51988DC0238.

- 11 Matteo Ciucci, "Internal Energy Market," European Parliament, April 2024, https://www.europarl. europa.eu/factsheets/en/sheet/45/internal-energy-market.
- 12 Dutton, "EU Energy Policy and the Third Package."
- 13 European Commission, *Communication from the Commission: Clean Energy for All Europeans*, COM/2016/0860 (Brussels: European Commission, November 2016), https://eur-lex.europa.eu/legalcontent/en/TXT/?uri=CELEX:52016DC0860.
- 14 "Shrinking Market Shares for Largest EU Energy Companies," Eurostat, March 14, 2023, https://ec.europa.eu/eurostat/web/products-eurostat-news/w/DDN-20230314-1.
- 15 Isabel Muttreja and Bernhard Blumenau, "How Russia's Invasion Changed German Foreign Policy," Chatham House, November 30, 2022, https://www.chathamhouse.org/2022/11/how-russias-invasionchanged-german-foreign-policy.
- 16 Daniel Yergin, *The New Map: Energy, Climate, and the Clash of Nations* (New York: Penguin, 2021), 75-76.
- 17 C. J. Chivers, "Gazprom Refuses to Back Down on Ukraine," *New York Times*, December 31, 2005, https://www.nytimes.com/2005/12/31/world/europe/gazprom-refuses-to-back-down-on-ukraine.html.
- 18 C. J. Chivers, "Explosions in Russia Cut Gas Pipelines to Georgia," *New York Times*, January 22, 2006, https://www.nytimes.com/2006/01/22/international/europe/explosions-in-russia-cut-gas-pipelines-togeorgia.html.
- 19 Gabriel Collins, "Russia's Use of the 'Energy Weapon' in Europe," Rice University's Baker Institute for Public Policy, July 18, 2017, https://www.bakerinstitute.org/research/russias-use-energy-weapon-europe.
- 20 Eurostat, "Energy Statistics–An Overview," European Commission, May 27, 2024, https://ec.europa.eu/ eurostat/statistics-explained/index.php?title=Energy_statistics__an_overview.
- 21 Szymon Kardaś, "Conscious Uncoupling: Europeans' Russian Gas Challenge in 2023," European Council on Foreign Relations, February 13, 2023, https://ecfr.eu/article/conscious-uncoupling-europeans-russiangas-challenge-in-2023/.
- 22 "Frequently Asked Questions on Energy Security," IEA, November 16, 2022, https://www.iea.org/articles/ frequently-asked-questions-on-energy-security.
- 23 Martin Russell, *The Nord Stream 2 Pipeline: Economic, Environmental and Geopolitical Issues* (Brussels: European Parliamentary Research Service, July 2022), https://www.europarl.europa.eu/RegData/etudes/ BRIE/2021/690705/EPRS_BRI(2021)690705_EN.pdf.
- 24 Matthew Karnitschnig, "Germany Revives Putin's Pipeline Dream," *Politico*, December 13, 2017, https:// www.politico.eu/article/germany-revives-putins-pipeline-dream/; and Julian Wettengel, "Nord Stream 2–Symbol of Failed German Bet on Russian Gas," Clean Energy Wire, March 8, 2023, https://www. cleanenergywire.org/factsheets/gas-pipeline-nord-stream-2-links-germany-russia-splits-europe.
- 25 "US Government Increases Pressure on Germany to Halt Nord Stream 2," Clean Energy Wire, December 18, 2017, https://www.cleanenergywire.org/news/spd-should-focus-less-climate-foreign-minpower-price-2017/us-government-increases-pressure-germany-halt-nord-stream-2; "Environmental Organisations Urge German Parties to Scrap Nord Stream 2 Pipeline," Clean Energy Wire, January 12, 2018, https://www.cleanenergywire.org/news/breakthrough-german-coalition-talks-energiewendecosts-unknown/environmental-organisations-urge-german-parties-scrap-nord-stream-2-pipeline; "Cross-Party German Alliance Opposes Nord Stream 2 Pipeline," Clean Energy Wire, February 21, 2018, https://www.cleanenergywire.org/news/car-club-calls-diesel-retrofits-germans-split-over-diesel-bans/ cross-party-german-alliance-opposes-nord-stream-2-pipeline; and "Ukrainian President Says Russian-German Gas Pipeline Will 'Feed the Beast' in Moscow," Clean Energy Wire, May 18, 2018, https://www.

cleanenergywire.org/news/reactions-eu-air-quality-lawsuit-100-renewables-possible-study/ukrainian-president-says-russian-german-gas-pipeline-will-feed-beast-moscow.

- 26 Bergmann et al., "Energy Security and Geopolitics Conference."
- 27 Ibid.
- Alan Riley, "Can Nord Stream 2 Really Replace Groningen?," *American Interest*, October 2, 2018, https://www.the-american-interest.com/2018/10/02/can-nord-stream-2-really-replace-groningen/.
- 29 Casey Crownhart, "Inside Germany's Power Struggle over Nuclear Energy," *MIT Technology Review*, April 27, 2023, https://www.technologyreview.com/2023/04/27/1072287/inside-germanys-power-struggle-over-nuclear-energy/.
- 30 Christoph Hasselbach, "Fukushima Triggered Germany's Nuclear Phaseout," DW, March 10, 2021, https://www.dw.com/en/how-fukushima-triggered-germanys-nuclear-phaseout/a-56829217.
- 31 Crownhart, "Inside Germany's Power Struggle."
- 32 Lisa Jucca, "Gazprom Sale Fumble Triggers Canny German Pounce," Reuters, April 5, 2022, https://www. reuters.com/breakingviews/gazprom-sale-fumble-triggers-canny-german-pounce-2022-04-05/.
- 33 "Nord Stream 2: Trump Approves Sanctions on Russia Gas Pipeline," BBC, December 21, 2019, https:// www.bbc.com/news/world-europe-50875935.
- ³⁴ "Fact Sheet on U.S. Opposition to Nord Stream 2," U.S. Department of State, December 27, 2019, https://2017-2021.state.gov/fact-sheet-on-u-s-opposition-to-nord-stream-2/.
- 35 Timothy Rooks, "What's behind America's Nord Stream Objections?," DW, February 7, 2019, https://www.dw.com/en/whats-behind-americas-nord-stream-objections/a-47407171.
- 36 Patrick Wintour, "US Warns Nord Stream Pipeline Is 'Russian Geopolitical Project'," *The Guardian*, March 18, 2021, https://www.theguardian.com/world/2021/mar/18/us-warns-nord-stream-pipeline-isrussian-geopolitical-project.
- 37 "Remarks by President Biden and Chancellor Merkel of the Federal Republic of Germany in Press Conference," The White House, July 15, 2021, https://www.whitehouse.gov/briefing-room/speechesremarks/2021/07/15/remarks-by-president-biden-and-chancellor-merkel-of-the-federal-republic-ofgermany-in-press-conference/.
- 38 Ibid.
- 39 Kiron Skinner, "Biden's Surrender to Merkel on Nord Stream 2," Foreign Policy, July 26, 2021, https:// foreignpolicy.com/2021/07/26/biden-merkel-putin-nord-stream-2-gas-pipeline-germany-russia-ukrainegeopolitics-central-eastern-europe/.
- 40 Vladimir Soldatkin, "Russia Completes Nord Stream 2 Construction, Gas Flows Yet to Start," Reuters, September 10, 2021, https://www.reuters.com/business/energy/russias-gazprom-says-it-has-completednord-stream-2-construction-2021-09-10/.
- 41 Zosia Wanat, "Poland Finds It's Tough to Go Cold Turkey on Russian Energy," *Politico*, April 25, 2022, https://www.politico.eu/article/poland-russia-energy-oil-gas-coal-ukraine-war-sanctions-prime-ministermateusz-morawiecki/.
- 42 Directorate-General for Energy, *RePowerEU: One Year Later–Latvia* (Brussels: European Commission, May 2023), https://energy.ec.europa.eu/document/download/4124469f-ffla-4b8f-8813-44096d2fb8fc_ en?filename=LV_REPowerEU.pdf.
- 43 Benjamin Moll, Moritz Schularick, and Georg Zachmann, "The Power of Substitution: The Great German

Gas Debate in Retrospect," Brookings Institution, September 23, 2023, https://www.brookings.edu/articles/the-power-of-substitution-the-great-german-gas-debate-in-retrospect/.

- 44 "The European Green Deal," European Commission, accessed June 24, 2024, https://commission. europa.eu/strategy-and-policy/priorities-2019-2024/european-green-deal_en.
- 45 Ibid.
- 46 European Commission, *Fit for 55*.
- 47 Emily Benson et al., "Analyzing the European Union's Carbon Border Adjustment Mechanism," CSIS, CSIS Brief, February 17, 2023, https://www.csis.org/analysis/analyzing-european-unions-carbon-borderadjustment-mechanism; and ETS News, "EU Adopts Landmark ETS Reforms and New Policies to Meet 2030 Target," International Carbon Action Partnership, May 3, 2023, https://icapcarbonaction.com/en/ news/eu-adopts-landmark-ets-reforms-and-new-policies-meet-2030-target.
- 48 "EU Taxonomy for Sustainable Activities," European Commission, https://finance.ec.europa.eu/ sustainable-finance/tools-and-standards/eu-taxonomy-sustainable-activities_en.
- 49 "The European Green Deal Investment Plan and Just Transition Mechanism Explained," European Commission, January 14, 2020, https://ec.europa.eu/commission/presscorner/detail/en/qanda_20_24.
- 50 Camilla Naschert, "'It's About Trust': EU Green Taxonomy Scrutinized over Inclusion of Nuclear, Gas," S&P Global, January 5, 2022, https://www.spglobal.com/marketintelligence/en/news-insights/latestnews-headlines/it-s-about-trust-eu-green-taxonomy-scrutinized-over-inclusion-of-nuclear-gas-68279835.
- 51 Hans von der Burchard, "Germany to Vote against EU Plan to Label Nuclear as Green but Won't Sue," *Politico*, May 13, 2022, https://www.politico.eu/article/germany-to-veto-eu-plan-to-label-nuclear-energyas-green-european-commission/.
- 52 Kate Abnett, "EU Parliament Backs Labelling Gas and Nuclear Investments as Green," Reuters, July 6, 2022, https://www.reuters.com/business/sustainable-business/eu-parliament-vote-green-gas-nuclear-rules-2022-07-06/.
- 53 von der Burchard, "Germany to Vote."
- 54 Ibid.
- 55 Rosie Frost, "Europe Is Divided on Nuclear Power: Which Countries Are for and against It?," Euronews, April 1, 2024, https://www.euronews.com/green/2024/04/01/europe-is-divided-on-nuclear-power-whichcountries-are-for-and-against-it.
- 56 Kerstine Appunn, "The History behind Germany's Nuclear Phase-out," Clean Energy Wire, March 9, 2021, https://www.cleanenergywire.org/factsheets/history-behind-germanys-nuclear-phase-out.
- 57 "Nuclear Power in France," World Nuclear Association, updated May 21, 2024, https://world-nuclear.org/ information-library/country-profiles/countries-a-f/france.
- 58 "Europe's Coal Exit," Beyond Fossil Fuels, June 10, 2024, https://beyondfossilfuels.org/europes-coalexit/.
- 59 Bate Felix, "France Sets 2050 Carbon-Neutral Target with New Law," Reuters, June 27, 2019, https:// www.reuters.com/article/us-france-energy/france-sets-2050-carbon-neutral-target-with-new-lawidUSKCNITS30B/; and John Upton, "E.U. Loophole Counts Wood Energy as 'Carbon Neutral.' It's Not." Grist, January 1, 2017, https://grist.org/culture/e-u-loophole-counts-wood-energy-as-carbon-neutral-itsnot/.
- 60 Benjamin Wehrmann, "Experts Criticise Proposed German Coal Exit Law for Deviating from

Commission Compromise," Clean Energy Wire, May 25, 2020, https://www.cleanenergywire.org/news/experts-criticise-proposed-german-coal-exit-law-deviating-commission-compromise.

- 61 Kate Connolly, "'Historic' German Ruling Says Climate Goals Not Tough Enough," *The Guardian*, April 29, 2021, https://www.theguardian.com/world/2021/apr/29/historic-german-ruling-says-climate-goals-not-tough-enough.
- 62 Reuters, "Germany's SPD, Greens, FDP Agree on Coal Exit by 2030," Reuters, October 15, 2021, https:// www.reuters.com/world/europe/germanys-spd-greens-fdp-agree-coal-exit-by-2030-2021-10-15/.
- 63 "На Ямале произошел пожар на заводе 'Газпром переработки'" [There was a fire at the "Gazprom Pererabotka" plant in Yamal], Interfax, August 5, 2021, https://www.interfax.ru/russia/782657.
- 64 "In September 2021, Gas Transit through Ukraine Exceeded 3 Billion Cubic Meters," UA Transmission System Operator, October 7, 2021, https://tsoua.com/en/news/in-september-2021-gas-transit-throughukraine-exceeded-3-billion-cubic-meters/.
- 65 Ignacio Urbasos, *The Future of Russian Gas in the EU* (Madrid: Real Instituto Elcano, March 2024), https:// media.realinstitutoelcano.org/wp-content/uploads/2024/03/policy-paper-the-future-of-russian-gas-inthe-eu.pdf.
- 66 Sam Meredith, "Russia Is Pumping A Lot Less Natural Gas to Europe All of a Sudden–and It Is Not Clear Why," CNBC, August 24, 2021, https://www.cnbc.com/2021/08/24/russia-is-pumping-less-natural-gas-toeurope-as-nord-stream-2-nears-completion.html.
- 67 Ibid.
- 68 Hans von der Burchard, "Only 4 Months before War, Germany Claimed Russian Gas Pipeline Posed No Risk," Politico, October 13, 2022, https://www.politico.eu/article/germany-release-merkel-eraassessment-saying-nord-stream-russia-gas-pose-no-risk/; and Bundesministerium für Wirtschaft und Energie [Federal Ministry for Economic Affairs and Energy], *Versorgungssicherheitsprüfung gem. § 4b Abs. 2 und Abs. 3 EnWG [Security of Supply Review according to § 4b para. 2 and para. 3 EnWG*] (Berlin: Bundesministerium für Wirtschaft und Energie, October 2021), https://www.politico.eu/wp-content/ uploads/2022/10/13/211026-prufung-versorgungssicherheit-nord-stream-2-bmwi.pdf.
- 69 Bundesministerium für Wirtschaft und Energie, Versorgungssicherheitsprüfung.
- 70 Vitaly Yermakov, *Big Bounce: Russian Gas amid Market Tightness* (Oxford: Oxford Institute for Energy Studies, September 2021), https://www.oxfordenergy.org/wpcms/wp-content/uploads/2021/09/Russian-gas-amid-market-tightness.pdf.
- 71 Julian E. Barnes and Eric Schmitt, "U.S. Warns Allies of Possible Russian Incursion as Troops Amass Near Ukraine," *New York Times*, November 19, 2021, https://www.nytimes.com/2021/11/19/us/politics/russia-ukraine-biden-administration.html.
- 72 "Policy Statement by Olaf Scholz, Chancellor of the Federal Republic of Germany and Member of the German Bundestag, 27 February 2022 in Berlin," German Federal Government, February 27, 2022, https://www.bundesregierung.de/breg-en/news/policy-statement-by-olaf-scholz-chancellor-of-thefederal-republic-of-germany-and-member-of-the-german-bundestag-27-february-2022-in-berlin-2008378.
- 73 "Executive Order on Blocking Property of Certain Persons and Prohibiting Certain Transactions with Respect to Continued Russian Efforts to Undermine the Sovereignty and Territorial Integrity of Ukraine," The White House, February 21, 2022, https://www.whitehouse.gov/briefing-room/ presidential-actions/2022/02/21/executive-order-on-blocking-property-of-certain-persons-andprohibiting-certain-transactions-with-respect-to-continued-russian-efforts-to-undermine-thesovereignty-and-territorial-integrity-of-ukraine/; "Council Implementing Regulation (EU) 2022/336

of 28 February 2022 Implementing Regulation (EU) No 269/2014 Concerning Restrictive Measures in Respect of Actions Undermining or Threatening the Territorial Integrity, Sovereignty and Independence of Ukraine," European Union Law, February 28, 2022, https://eur-lex.europa.eu/ legal-content/EN/TXT/?uri=CELEX:32022R0336; and Scott Horsley, "In an Effort to Choke Russian Economy, New Sanctions Target Russia's Central Bank," NPR, February 28, 2022, https://www.npr. org/2022/02/28/1083580974/in-an-effort-to-choke-russian-economy-new-sanctions-target-russiascentral-bank?t=1646164495488.

- 74 "Allies Freeze \$330 Bn of Russian Assets since Ukraine Invasion: Task Force," France24, June 29, 2022, https://www.france24.com/en/live-news/20220629-allies-freeze-330-bn-of-russian-assets-since-ukraine-invasion-task-force.
- 75 "Inside Europe's Energy Crisis," S&P Global, updated March 25, 2022, https://www.spglobal.com/en/ research-insights/special-reports/europe-energy-crisis.
- 76 "Dutch PM: Not Possible for Europe to Cut Off Russian Oil and Gas Immediately," Reuters, March 9, 2022, https://www.reuters.com/business/energy/dutch-pm-not-possible-europe-cut-off-russian-oil-gas-immediately-2022-03-09/?ref=hir.harvard.edu.
- 77 Hans von der Burchard and Merlin Sugue, "Germany Sticks with Putin for Its Oil and Gas," *Politico*, March 7, 2022, https://www.politico.eu/article/germany-putin-oil-gas/.
- 78 Ibid.
- 79 Ibid.
- 80 Ben McWilliams, Simone Tagliapietra, and Georg Zachmann, "The Letta Report: An Assessment of the Energy Proposals," Bruegel, April 19, 2024, https://www.bruegel.org/first-glance/letta-report-assessmentenergy-proposals.
- 81 Anna Cooban, "Germany Faces \$240 Billion Hit If Russian Gas Is Cut Off," CNN, April 13, 2024, https:// www.cnn.com/2022/04/13/economy/german-economy-russian-gas/index.html.
- 82 Moll, Schularick, and Zachmann, "The Power of Substitution."
- 83 Karen Pittel et al., "What If Germany Is Cut Off from Russian Energy?," Centre for Economic Policy Research, March 25, 2022, https://cepr.org/voxeu/columns/what-if-germany-cut-russian-energy.
- 84 Moll, Schularick, and Zachmann, "The Power of Substitution."
- 85 Jakob Feveile Adolfsen et al., "Oil Price Developments and Russian Oil Flows since the EU Embargo and G7 Price Cap," European Central Bank, February, 2023, https://www.ecb.europa.eu/press/economicbulletin/focus/2023/html/ecb.ebbox202302_02-59c965249a.en.html; Victor Jack, "Bulgaria Accelerates End of Sanctions Loophole That Earned Russia €1B," *Politico*, November 17, 2023, https://www.politico. eu/article/bulgaria-accelerates-end-of-sanctions-loophole-that-earned-russia-e1b/; and Kate Abnett, Jan Strupczewski, and Ingrid Melander, "EU Agrees Russia Oil Embargo, Gives Hungary Exemptions; Zelenskiy Vows More Sanctions," Reuters, March 31, 2022, https://www.reuters.com/world/europe/bestwe-could-get-eu-bows-hungarian-demands-agree-russian-oil-ban-2022-05-31/.
- 86 Council of the European Union, "Russia's Military Aggression against Ukraine: EU Imposes Sanctions against President Putin and Foreign Minister Lavrov and Adopts Wide Ranging Individual and Economic Sanctions," Press release, February 22, 2022, https://www.consilium.europa.eu/en/press/ press-releases/2022/02/25/russia-s-military-aggression-against-ukraine-eu-imposes-sanctions-againstpresident-putin-and-foreign-minister-lavrov-and-adopts-wide-ranging-individual-and-economicsanctions/; and "EU Adopts Package of Sanctions in Response to Russian Recognition of the Nongovernment Controlled Areas of the Donetsk and Luhansk Oblasts of Ukraine and Sending of Troops

into the Region," Council of the European Union, February 23, 2022, https://www.consilium.europa.eu/en/press/press-releases/2022/02/23/russian-recognition-of-the-non-government-controlled-areas-of-the-donetsk-and-luhansk-oblasts-of-ukraine-as-independent-entities-eu-adopts-package-of-sanctions/.

- 87 Joseph Nasr and Mark Trevelyan, "Putin Tells Europe: Pay in Roubles or We'll Cut Off Your Gas," Reuters, March 31, 2022, https://www.reuters.com/business/energy/russia-sets-deadline-rouble-gas-paymentseurope-calls-it-blackmail-2022-03-31/.
- 88 Remina Aleksieva and Kalina Tcolova, *Energy without Russia: The Consequences of the Ukraine War and the EU Sanctions on the Energy Sector in Europe–Country Report Bulgaria* (Budapest: Friedrich-Ebert-Stiftung, 2023), https://library.fes.de/pdf-files/bueros/budapest/20559.pdf.
- 89 Benjamin Wehrmann, "Gazprom Announcement to Reduce Nord Stream Gas Flows 'Political Decision'– Econ Min," Clean Energy Wire, June 15, 2022, https://www.cleanenergywire.org/news/gazpromannouncement-reduce-nord-stream-gas-flows-could-jeopardise-german-storage-target.
- 90 Associated Press, "Macron Warns France to Brace for Russian Gas Cutoff," PBS News Hour, July 14, 2022, https://www.pbs.org/newshour/world/macron-warns-france-to-brace-for-russian-gas-cutoff.
- 91 Laurenz Gehrke, Karl Mathiesen, and Paola Tamma, "Russia Declares Gas War on EU," Politico, June 20, 2022, https://www.politico.eu/article/russia-declares-gas-war-on-eu/.
- 92 "Who Blew up the Nord Stream Pipelines?," *The Economist*, March 8, 2023, https://www.economist.com/ the-economist-explains/2023/03/08/who-blew-up-the-nord-stream-pipelines.
- 93 "New Intelligence Points to Pro-Ukraine Group in Nord Stream Attack–NYT," Reuters, March 8, 2023, https://www.reuters.com/world/europe/us-intelligence-suggests-pro-ukrainian-group-sabotaged-nordstream-pipelines-nyt-2023-03-07/.
- 94 Adam Entous, Julian E. Barnes, and Adam Goldman, "Intelligence Suggests Pro-Ukrainian Group Sabotaged Pipelines, U.S. Officials Say," *New York Times*, updated March 8, 2023, https://www.nytimes. com/2023/03/07/us/politics/nord-stream-pipeline-sabotage-ukraine.html.
- 95 Stine Jacobsen and Lousie Rasmussen, "Denmark Ends Probe into 'Deliberate' Nord Stream Pipeline Blasts," Reuters, February 26, 2024, https://www.reuters.com/world/europe/denmark-endsinvestigation-into-nord-stream-pipeline-blasts-2024-02-26/; and Johan Ahlander and Anna Ringstrom, "Sweden Ends Nord Stream Sabotage Probe, Hands Evidence to Germany," Reuters, February 7, 2024, https://www.reuters.com/world/europe/sweden-ends-investigation-into-nord-stream-pipelineblasts-2024-02-07/.
- 96 Andrea Toreti et al., *Drought in Europe August 2022*, EUR 31192 EN (Luxembourg: Publications Office of the European Union, 2022), https://doi.org/10.2760/264241; and Philip Blenkinsop, "Europe Facing Its Worst Drought in 500 Years–Study," Reuters, August 23, 2022, https://www.reuters.com/world/europe/ nearly-two-thirds-europe-facing-drought-or-drought-risk-2022-08-23/.
- 97 Vera Eckert, "Low Rhine Water Level to Hit Output at Two German Coal Plants," Reuters, August 4, 2022, https://www.reuters.com/business/energy/low-rhine-water-level-hit-output-staudinger-5-coal-plant-2022-08-04/.
- 98 "France Tweaks Rules to Keep Nuclear Plants Running during Heatwave," Reuters, August 8, 2022, https://www.reuters.com/world/europe/frances-asn-nuclear-regulator-adapts-hot-water-discharge-ruleslight-heatwave-2022-08-08/.
- 99 "Nuclear Power in France," World Nuclear Association.
- 100 Ibid.

- 101 Autorité de Sûreté Nucléaire [Nuclear Safety Authority], *ASN Report: On the State of Nuclear Safety and Radiation Protection in France in 2022* (Montrouge, France: ASN, March 2023), 12, https://www.french-nuclear-safety.fr/annual_report/2022gb/12/#zoom=true.
- 102 "Carbon Price Tracker," Ember, accessed May 26, 2024, https://ember-climate.org/data/data-tools/ carbon-price-viewer/.
- 103 "Greece Energy Report," Enerdata, August 1, 2023, https://www.enerdata.net/estore/country-profiles/ greece.html.
- 104 Manuel Hidalgo Pérez et al., "The Effects of the Spanish Gas Cap on Prices, Inflation, and Consumption Six Months Later," EsadeEcPol, February 2, 2023, https://www.esade.edu/ecpol/en/publications/theeffects-of-the-spanish-gas-cap-on-prices-inflation-and-consumption-six-months-later/.
- 105 Annabelle Timsit and Emily Rauhala, "The E.U. Is Preparing for Blackouts This Winter, amid an Energy Crisis," *Washington Post*, October 5, 2022, https://www.washingtonpost.com/world/2022/10/05/europe-blackouts-energy-crisis-ukraine-russia/.
- 106 Marwa Rashad, "Europe's Ability to Survive a Cold Winter Hasn't Been Tested, Energy Execs Say," Reuters, April 23, 2024, https://www.reuters.com/world/europe/europes-ability-survive-cold-winterhasnt-been-tested-energy-execs-say-2024-04-23/.
- 107 "Where Does the EU's Gas Come From?," Council of the European Union, March 21, 2024, https://www.consilium.europa.eu/en/infographics/eu-gas-supply/#0.
- 108 Sarah Brown, "Coal Is Not Making a Comeback: Europe Plans Limited Increase," Ember, July 13, 2022, https://ember-climate.org/insights/research/coal-is-not-making-a-comeback/.
- 109 Pritish Raj and Sarah Matthews, "European Thermal Coal Buyers Eye Reselling in Asia amid Sufficient Stockpile," S&P Global, May 24, 2023, https://www.spglobal.com/commodityinsights/en/market-insights/ latest-news/coal/052423-european-thermal-coal-buyers-eye-reselling-in-asia-amid-sufficient-stockpile.
- 110 "Belgium to Extend Life of Two Nuclear Reactors by 10 Years," Reuters, January 9, 2023, https://www.reuters.com/business/energy/belgium-extend-life-two-nuclear-reactors-by-10-years-2023-01-09/.
- 111 "The Debate on a Lifetime Extension for Nuclear Power Plants," Federal Office for the Safety of Nuclear Waste Management, April 19, 2023, https://www.base.bund.de/EN/ns/nuclear-phase-out/lifetime-extension-npp-faq.html.
- 112 "Nuclear Power in France," World Nuclear Association.
- 113 Alfie Shaw, "France Sees Potential for 14 New Nuclear Reactors," Power Technology, January 8, 2024, https://www.power-technology.com/news/france-may-build-14-new-nuclear-reactors/.
- 114 "European Nuclear Alliance Leader's Declaration," *Élysée*, March 21, 2024, https://www.elysee.fr/en/ emmanuel-macron/2024/03/21/european-nuclear-alliance-leaders-declaration.
- 115 Frost, "Europe Is Divided."
- 116 Associated Press, "Plans for Poland's First Nuclear Power Plant Move Ahead as US and Polish Officials Sign an Agreement," AP News, September 27, 2023, https://apnews.com/article/poland-us-nuclearenergy-power-plant-deal-6905023bc532240ec06baaa1c205e93c.
- 117 "South Korea and Poland Enhance Cooperation in Nuclear," World Nuclear News, July 14, 2023, https://world-nuclear-news.org/Articles/South-Korea-and-Poland-enhance-cooperation-in-nucl.
- 118 Associated Press, "Czech Government Now Seeks to Build up to 4 Nuclear Reactors Instead 1 to Reduce Price," AP News, January 21, 2024, https://apnews.com/article/czech-nuclear-reactors-fossil-fuels-energy-

fafe34fb2e16446a75eeca7fddfe5e02.

- 119 Stanley Reed, "With Russia's Exit, Norway Becomes Europe's Energy Champion," New York Times, April 6, 2023, https://www.nytimes.com/2023/04/06/business/energy-environment/ukraine-russia-wareurope-energy.html?auth=login-google1tap&login=google1tap.
- 120 "New Delivery Records for Norwegian Natural Gas," Gassco, January 10, 2024, https://gassco.eu/en/new-delivery-records-for-norwegian-natural-gas/.
- 121 Ibid.
- 122 Priscila Azevedo Rocha and Kari Lundgren, "Europe's Gas Supply Once Again Hinges on One Company," Bloomberg, May 13, 2024, https://www.bloomberg.com/news/features/2024-05-13/equinor-how-norways-gas-giant-quietly-took-over-europe?sref=B2BBHw9t.
- 123 Malte Humpert, "Norway Supplied Almost Half of German Natural Gas Imports during First Nine Months of 2023," High North News, October 3, 2023, https://www.highnorthnews.com/en/norway-supplied-almost-half-german-natural-gas-imports-during-first-nine-months-2023.
- 124 "Norway Is Profiting Embarrassingly from War in Europe," *The Economist*, September 8, 2022, https://www.economist.com/europe/2022/09/08/norway-is-profiting-embarrassingly-from-war-in-europe.
- 125 Rocha and Lundgren, "Europe's Gas Supply."
- 126 European Commission, *Quarterly Report on European Gas Markets* (Brussels, Belgium: European Commission, 2023), vol. 16, no. 2, https://energy.ec.europa.eu/system/files/2023-12/New_Quarterly_Report_on_European_Gas_markets_Q2_2023.pdf.
- 127 Arnold C. Dupuy, "A New Black Sea Natural Gas Project Could Be a Game Changer for the Region–and a Challenge for Putin," *TURKEYSource* (blog), Atlantic Council, July 26, 2023, https://www.atlanticcouncil. org/blogs/turkeysource/a-new-black-sea-natural-gas-project-could-be-a-game-changer-for-the-regionand-a-challenge-for-putin/.
- 128 "Tyra, a State-of-the-Art Offshore Gas Hub in the North Sea," TotalEnergies, https://totalenergies.com/ projects/gas/tyra-state-art-offshore-gas-hub-north-sea.
- 129 George Zachmann et al., "European Natural Gas Imports," Bruegel, June 5, 2024, https://www.bruegel. org/dataset/european-natural-gas-imports.
- 130 "Joint Statement by President Biden and President von der Leyen on European Energy Security," The White House, June 27, 2022, https://www.whitehouse.gov/briefing-room/statementsreleases/2022/06/27/joint-statement-by-president-biden-and-president-von-der-leyen-on-europeanenergy-security/.
- 131 "Assistant Secretary Geoffrey Pyatt North American Gas Forum," U.S. Department of State, October 24, 2023, https://www.state.gov/assistant-secretary-geoffrey-pyatt-north-american-gas-forum/.
- 132 Yuka Obayashi and Marwa Rashad, "Japan to Divert LNG to Europe amid Russia-Ukraine Tension," Reuters, February 9, 2022, https://www.reuters.com/business/energy/japan-diverting-lng-europe-somealready-route-industry-minister-2022-02-09/.
- 133 "Natural Gas Weekly Update," U.S. Energy Information Administration, January 26, 2023, https://www.eia.gov/naturalgas/weekly/archivenew_ngwu/2023/01_26/.
- 134 "Breaking Down US LNG Contract Value," Timera Energy, June 14, 2021, https://timera-energy.com/blog/ breaking-down-us-lng-contract-value/.
- 135 International Energy Agency, Global Gas Security Review 2023 (Paris: IEA, 2023),

https://iea.blob.core.windows.net/assets/f45a2340-8479-4585-b26e-ec5e9b14feca/ GlobalGasSecurityReview2023IncludingtheGasMarketReportQ32023.pdf.

- 136 "China's Natural Gas Consumption and LNG Imports Declined in 2022, amid Zero-COVID Policies," U.S. Energy Information Administration, June 1, 2023, https://www.eia.gov/todayinenergy/detail. php?id=56680.
- 137 Erwida Maulia, "LNG 'Inequality' Bites as Europe Takes Supply from Asia," Nikkei Asia, June 27, 2023, https://asia.nikkei.com/Business/Energy/LNG-inequality-bites-as-Europe-takes-supply-from-Asia.
- 138 Stephen Stapczynski, "Pakistan Faces Deeper Power Crisis as LNG Becomes Too Expensive," Bloomberg, June 24, 2022, https://www.bloomberg.com/news/articles/2022-06-24/pakistan-faces-deeper-powercrisis-as-lng-becomes-too-expensive?sref=B2BBHw9t.
- 139 "The Influence of New LNG Terminals on the Future EU Energy Market," Council of European Energy Regulators, February 16, 2024, https://www.ceer.eu/documents/104400/-/-/37ca015e-86dd-7a9d-c8d4-6b6b8e408d5f.
- 140 "Germany Builds Up LNG Import Terminals," Reuters, June 22, 2023, https://www.reuters.com/markets/ commodities/germanys-lng-import-project-plans-2023-02-15/.
- 141 EU Agency for the Cooperation of Energy Regulators, *Analysis of the European LNG Market Developments:* 2024 Market Monitoring Report (Ljubljana, Slovenia: ACER, April 2024), https://www.acer.europa.eu/ sites/default/files/documents/Publications/ACER_2024_MMR_European_LNG_market_developments.pdf.
- 142 Patrick Heather, "A Hub for Europe': The Iberian Promise?," Oxford Institute for Energy Studies, April 30, 2019, https://www.oxfordenergy.org/publications/hub-europe-iberian-promise/.
- 143 Zachmann et al., "European Natural Gas Imports."
- 144 "European LNG Tracker," IEEFA, February 2024, https://ieefa.org/european-lng-tracker.
- 145 ACER, Analysis of the European LNG Market Developments.
- 146 Michał Kędzierski, "An Abundance of Gas Ports. The Emergency Diversification of Gas Supplies in Germany," OSW Centre for Eastern Studies, May 20, 2022, https://www.osw.waw.pl/en/publikacje/oswcommentary/2022-05-20/abundance-gas-ports-emergency-diversification-gas-supplies.
- 147 Ben Cahill and Leslie Palti-Guzman, "The Role of Gas in Ukraine's Energy Future," CSIS, *CSIS Briefs*, January 13, 2023, https://www.csis.org/analysis/role-gas-ukraines-energy-future.
- 148 Max Bearak and Hilary Swift, "Why Greece Is Betting Big on American Gas," *New York Times*, May 31, 2024, sec. Climate, https://www.nytimes.com/2024/05/31/climate/greece-europe-natural-gas-lng.html.
- 149 Francesca Landini and Angelo Amante, "Exclusive: Italy in Talks with Bavaria on Gas and Hydrogen Exports, Eyes Eastern Europe," Reuters, September 27, 2023, https://www.reuters.com/markets/ emerging/italy-talks-with-bavaria-gas-hydrogen-exports-eyes-eastern-europe-2023-09-27/.
- 150 Ben McWilliams and Georg Zachmann, "European Natural Gas Demand Tracker," Bruegel, May 29, 2024, https://www.bruegel.org/dataset/european-natural-gas-demand-tracker.
- 151 Nicolas Garriga and Barbara Surk, "Eiffel Tower to Go Dark Earlier as Paris Saves Energy," AP News, September 13, 2022, https://apnews.com/article/russia-ukraine-france-paris-103898b33a6e8382f469eaac 5947f09e.
- 152 Sarah Brown and Dave Jones, "European Electricity Review 2024," Ember, February 7, 2024, https://ember-climate.org/insights/research/european-electricity-review-2024/.
- 153 McWilliams and Zachmann, "European Natural Gas Demand Tracker."

- 154 Anouk Honoré, *European Gas Demand Fundamentals:* Q3 2023 *Review, October 2023 First Glance, and Winter 2023/2024 Outlook* (Oxford: OIES, November 2023), https://www.oxfordenergy.org/wpcms/wpcontent/uploads/2023/11/Insight-138-European-gas-demand-fundamentals-.pdf.
- 155 Peter Zeniewski, Gergely Molnar, and Paul Hugues, "Europe's Energy Crisis: What Factors Drove the Record Fall in Natural Gas Demand in 2022?," IEA, March 14, 2023, https://www.iea.org/commentaries/ europe-s-energy-crisis-what-factors-drove-the-record-fall-in-natural-gas-demand-in-2022.
- 156 Dave Jones, Sarah Brown, and Paweł Czyżak, "European Electricity Review 2023," Ember, January 31, 2023, https://ember-climate.org/insights/research/european-electricity-review-2023/.
- 157 Anna Flávia Rochas, "EU Wind Targets under Threat from Volatile Costs," Reuters, November 25, 2022, https://www.reuters.com/business/energy/eu-wind-targets-under-threat-volatile-costs-2022-11-25/.
- 158 European Commission, Proposal for a Council Regulation Amending Regulation (EU) 2022/2577 Laying Down a Framework to Accelerate the Deployment of Renewable Energy (Brussels: European Commission, November 28, 2023), https://energy.ec.europa.eu/document/download/fa4522d8-7727-4879-856f-8569155953bc_en?filename=COM_2023_763_1_EN_ACT_part1_v8.pdf; and Martin Schachinger, "Solar Module Price Falling, with No End in Sight," pv magazine, September 25, 2023, https://www.pvmagazine.com/2023/09/25/solar-module-price-falling-with-no-end-in-sight/.
- 159 Bjarne Schieldrop, "German Solar Power Prices Are Collapsing as Market Hits Solar Saturation," SEB Research, May 21, 2024, https://research.sebgroup.com/macro-ficc/reports/49776.
- 160 Eamon Farhat, "Spanish Power Is Almost Free with Renewables Set for Record," Bloomberg, February 29, 2024, https://www.bloomberg.com/news/articles/2024-02-29/spanish-power-is-almost-free-with-renewables-set-for-record?sref=B2BBHw9t.
- 161 IEA, Renewables 2023: Analysis and Forecast to 2028 (Paris: IEA, January 2024), https://iea.blob.core. windows.net/assets/96d66a8b-d502-476b-ba94-54ffda84cf72/Renewables_2023.pdf; and ACER, *Analysis* of the European LNG Market Developments.
- 162 Federal Ministry for Economic Affairs and Climate Action, "First German LNG Terminal Inaugurated in Wilhelmshaven," Press release, December 17, 2022, https://www.bmwk.de/Redaktion/EN/Pressemitteilu ngen/2022/12/20221217-first-german-lng-terminal-inaugurated-in-wilhelmshaven.html.
- 163 Bergmann et al., "Energy Security and Geopolitics Conference."
- 164 Directorate-General for Energy, "Inauguration of Gas Interconnection between Poland and Lithuania," European Commission, May 5, 2022, https://commission.europa.eu/news/inauguration-gasinterconnection-between-poland-and-lithuania-2022-05-05_en.
- 165 "The Baltic Pipe: A Subsea Pipeline to Transport Natural Gas under the North Sea," European Commission, https://cinea.ec.europa.eu/featured-projects/baltic-pipe-subsea-pipeline-transport-naturalgas-under-north-sea_en.
- 166 Sgaravatti et al., "National Fiscal Policy."
- 167 Daniela Filip, Daphne Momferatou, and Ralph Setzer, "Inflation and Competitiveness Divergences in the Euro Area Countries," European Central Bank, *ECB Economic Bulletin*, no. 4, April 2023, https://www.ecb.europa.eu/press/economic-bulletin/focus/2023/html/ecb.ebbox202304_08-350974a969.en.html.
- 168 Moll, Schularick, and Zachmann, "The Power of Substitution."
- 169 Georgia Nakou, *Energy without Russia: The Consequences of the Ukraine War and the EU Sanctions on the Energy Sector in Europe–Country Report Greece* (Budapest: Friedrich-Ebert-Stiftung, 2023), https:// library.fes.de/pdf-files/bueros/budapest/20476.pdf.

- 170 Moll, Schularick, and Zachmann, "The Power of Substitution."
- 171 "Germany Completes Nationalization of Uniper–Ministries," Reuters, December 22, 2022, https://www.reuters.com/business/energy/germany-completes-nationalization-uniper-ministries-2022-12-22/.
- 172 Nikolaus J. Kurmayer, "Germany's Wounded Chemical Industry Causes Concern in Berlin," Euractiv, November 3, 2022, https://www.euractiv.com/section/economy-jobs/news/germanys-woundedchemical-industry-causes-concern-in-berlin/.
- 173 Giovanni Sgaravatti, Simone Tagliapietra, and Cecilia Trasi, "The Fiscal Side of Europe's Energy Crisis: The Facts, Problems and Prospects," Bruegel, March 2, 2023, https://www.bruegel.org/blog-post/fiscalside-europes-energy-crisis-facts-problems-and-prospects.
- 174 Filip, Momferatou, and Setzer, "Inflation and Competitiveness Divergences."
- 175 Melih Firat and Otso Hao, *Demand vs. Supply Decomposition of Inflation: Cross-Country Evidence with Applications* (Washington, DC: International Monetary Fund, October 2023), https://www.imf.org/en/ Publications/WP/Issues/2023/09/28/Demand-vs-539665.
- 176 European Commission, *Greece 2023 Country Report* (Brussels: European Commission, May 2023), https://economy-finance.ec.europa.eu/system/files/2023-05/EL_SWD_2023_608_en.pdf.
- 177 European Commission, Informal Meeting of the Heads of State or Government–Versailles Declaration (Brussels: European Commission, March 11, 2022), https://www.consilium.europa.eu/ media/54773/20220311-versailles-declaration-en.pdf.
- 178 Directorate-General for Energy, *RePowerEU*.
- 179 "EU Solar Energy Strategy," European Commission, https://ec.europa.eu/info/law/better-regulation/ have-your-say/initiatives/13338-EU-solar-energy-strategy_en.
- 180 "Gas Storage," European Commission, https://energy.ec.europa.eu/topics/energy-security/gas-storage_en.
- 181 "Security of Gas Supply," European Commission, https://energy.ec.europa.eu/topics/energy-security/ security-gas-supply_en.
- 182 Council of the European Union, "Security of Gas Supply: Member States Agree on Recommendation to Continue Voluntary Demand Reduction Measures," Press release, March 4, 2024, https://www. consilium.europa.eu/en/press/press-releases/2024/03/04/security-of-gas-supply-member-states-agreeon-recommendation-to-continue-voluntary-demand-reduction-measures/.
- 183 European Commission, "Energy Security: Commission Hosts First Meeting of EU Energy Purchase Platform to Secure Supply of Gas, LNG and Hydrogen," Press release, April 8, 2022, https://ec.europa.eu/ commission/presscorner/detail/es/ip_22_2387.
- 184 European Commission, "AggregateEU–One Year On," Press release, February 1, 2024, https://energy. ec.europa.eu/news/aggregateeu-one-year-2024-02-01_en.
- 185 Alex Barnes, *EU Joint Purchasing of Gas–An assessment*, OIES Paper NG 184 (Oxford: OIES, 2023), https:// www.oxfordenergy.org/wpcms/wp-content/uploads/2023/09/EU-Joint-Purchasing-of-Gas-NG184.pdf; and "Access to Infrastructure, Exemptions and Derogations," European Commission, updated January 30, 2024, https://energy.ec.europa.eu/topics/markets-and-consumers/wholesale-energy-market/accessinfrastructure-exemptions-and-derogations_en.
- 186 Council of the European Union, "Fit for 55: Council Signs Off on Gas and Hydrogen Market Package," Press release, May 21, 2024, https://www.consilium.europa.eu/en/press/press-releases/2024/05/21/ fit-for-55-council-signs-off-on-gas-and-hydrogen-market-package/; and Marta Pacheco, "EU Policy.

Commissioner Touts Launch of Critical Raw Material Joint Purchasing," Euronews, May 15, 2024, https://www.euronews.com/green/2024/05/15/commissioner-touts-launch-of-critical-raw-material-joint-purchasing.

- 187 European Commission, "Joint Statement between the European Commission and the United States on European Energy Security," Press release, March 25, 2022, https://ec.europa.eu/commission/ presscorner/detail/en/STATEMENT_22_2041.
- 188 "EU-U.S. Task Force on Energy Security: Progress Report and Outlook 2022-2023," European Commission, n.d., https://energy.ec.europa.eu/document/download/7446ed9f-1a74-46a8-9050-3971cde03876_en?filename=EU-US%20Energy%20Security%20TF_report_final_0.pdf.
- 189 European Association for Coal and Lignite, *Coal in Europe 2022* (Brussels: EURACOAL, April 2022), https://public.euracoal.eu/download/Public-Archive/Library/Charts-Maps/Coal-in-Europe/EURACOAL-Coal-in-Europe-2022-04.pdf.
- 190 Victor Jack, "Putin Rakes in Extra €1B for His War Chest via Bulgaria Sanctions Loophole," *Politico*, November 9, 2023, https://www.politico.eu/article/how-russia-made-e1b-from-an-eu-sanctions-loopholefor-bulgaria/.
- 191 Victor Jack, "Bulgaria Moves to End Russia Sanction Opt-Out as Pressure Mounts," *Politico*, December 13, 2023, https://www.politico.eu/article/bulgaria-end-russia-sanctions-opt-out-price-cap-loophole/.
- 192 Michał Kędzierski, "Kazakhstan Is Set to Supply Oil to Germany," OSW Centre for Eastern Studies, June 23, 2023, https://www.osw.waw.pl/en/publikacje/analyses/2023-06-23/kazakhstan-set-to-supply-oil-to-germany.
- 193 Simon Suzan and Agathe Bounfour, New Oil Map: Impact of Russia's War on Ukraine on Supply and Demand (Brussels: Transport & Environment, July 2023), https://www.transportenvironment.org/ uploads/files/202307_oil_imports_report_compressed.pdf; and Centre for Research on Energy and Clean Air, "The Laundromat: EU, G7 and Australia's Indirect Imports of Russian Oil Revealed," Press release, April 19, 2023, https://energyandcleanair.org/wp/wp-content/uploads/2023/04/CREA_Press-release_Thelaundromat_EU-G7-and-Australias-indirect-imports-of-Russian-oil-revealed.pdf.
- 194 Ben Cahill, "Big Challenges for Russian Oil Price Cap," CSIS, *Commentary*, September 9, 2022, https://www.csis.org/analysis/big-challenges-russian-oil-price-cap.
- 195 Ana Maria Jaller-Makarewicz, "EU Turns a Blind Eye to 21% of Russian LNG Flowing through Its Terminals," Institute for Energy Economics and Financial Analysis, November 29, 2023, https://ieefa.org/ resources/eu-turns-blind-eye-21-russian-lng-flowing-through-its-terminals.
- 196 McWilliams et al., "The European Union-Russia Energy Divorce."
- 197 "Europe Doubled Its Import of Russian Nuclear Fuel for 2023, Data Say," Bellona, March 15, 2024, https://bellona.org/news/nuclear-issues/2024-03-europe-russian-nuclear-fuel.
- 198 Ben McWilliams et al., "The EU Can Manage without Russian Liquified Natural Gas," Bruegel, June 28, 2023, https://www.bruegel.org/policy-brief/eu-can-manage-without-russian-liquified-natural-gas.
- 199 Euronews and Reuters, "Hungary Strikes Deal with Gazprom to Defer Payment for Gas–Foreign Minister," TASS, October 12, 2022, https://tass.com/economy/1521943; and "Russia's Gazprom to Increase Gas Supplies to Hungary," Euronews, October 23, 2023, https://www.euronews.com/ business/2023/10/23/russias-gazprom-to-increase-gas-supplies-to-hungary.
- 200 Monika Dulian, *Recast EU Regulation on Gas and Hydrogen Networks* (Strasbourg, France: European Parliamentary Research Service, February 2024), https://www.europarl.europa.eu/thinktank/en/ document/EPRS_BRI(2022)729302.

- 201 Cahill and Palti-Guzman, "The Role of Gas."
- 202 Urbasos, The Future of Russian Gas.
- ²⁰³ "Europe's benchmark natural gas prices have fallen massively from 2022." Robin Brooks, X post, March 15, 2024, 7:48 AM, https://x.com/robin_j_brooks/status/1768605088463888573.
- 204 European Commission, "European Consumers and Industry to Benefit from Clean, Secure and Stable Energy Supplies with Key Market Reforms Now Adopted," Press release, May 21, 2024, https://ec.europa. eu/commission/presscorner/detail/en/ip_24_2261.
- 205 "Electricity Market Reform," Council of the European Union, updated May 22, 2024, https://www. consilium.europa.eu/en/policies/electricity-market-reform/.
- 206 Ben McWilliams and Georg Zachmann, "Four Questions for Germany's Big Hydrogen Power Plan," Bruegel, February 19, 2024, https://www.bruegel.org/analysis/four-questions-germanys-big-hydrogenpower-plan.
- 207 Directorate-General for Energy, RePowerEU.
- 208 Ignacio Urbasos Arbeloa, "EU Hydrogen Diplomacy 2.0: Aligning Climate Ambition and Energy Security," Real Institute Elcano, October 17, 2023, https://www.realinstitutoelcano.org/en/analyses/euhydrogen-diplomacy-2-O-aligning-climate-ambition-and-energy-security/; and America Hernandez, "Go Big or Go Green? The EU's Massively Expanding Hydrogen Bet," *Politico*, October 19, 2022, https://www. politico.eu/article/go-big-or-go-green-the-eus-massively-expanding-hydrogen-bet/.
- 209 Strategy&, "Navigating the Global Hydrogen Ecosystem," PwC, https://cloud.email.strategyand.pwc.com/ content-form?qs=dc7fcb176550eb1a6d1e950e3a1ac396b5c08c8064100f9d35f19f4202242ac5b13c4f27ed3e 3170860bc13f5294f0630fcf34015246ca3b7d9ff81e86e4c36fda682d8ad5b44c1d057fa6e60848d26b.
- 210 Urbasos Arbeloa, "EU Hydrogen Diplomacy 2.0."
- 211 Polly Martin, "EU Hydrogen Targets Are 'Impossible' as Green H2 Costs Eight Times As Much As Grey Hydrogen: Total CEO," Hydrogen Insight, April 29, 2024, https://www.hydrogeninsight.com/production/ eu-hydrogen-targets-are-impossible-as-green-h2-costs-eight-times-as-much-as-grey-h2-today-totalceo/2-1-1634747.
- 212 Directorate-General for Energy, "Memorandum of Understanding between the European Union and Ukraine on a Strategic Partnership on Biomethane, Hydrogen and Other Synthetic Gases," European Commission, February 2, 2023, https://energy.ec.europa.eu/publications/memorandum-understandingbetween-european-union-and-ukraine-strategic-partnership-biomethane_en.
- 213 "EU Forges Hydrogen Partnerships and Disputes Policies," Hydrogen Technology Expo, November 12, 2022, https://www.hydrogen-expo.com/industry-news/eu-forges-hydrogen-partnerships-disputespolicies.
- 214 European Commission, "Commission Holds Exploratory Energy Transition Talks with Saudi Arabia," Press release, April 28, 2024, https://ec.europa.eu/commission/presscorner/detail/en/read_24_2341.
- 215 European Commission, "Making Further Progress in the Strategic Partnership at 2023 EU–Canada Summit," Press release, November 24, 2023, https://ec.europa.eu/commission/presscorner/detail/ en/IP_23_6018; and Ignacio Urbasos, "Green Hydrogen–Bringing Europe and Latin America Closer Together?," IPS, September 19, 2023, https://www.ips-journal.eu/topics/economy-and-ecology/greenhydrogen-bringing-europe-and-latin-america-closer-together-6994/.
- 216 "Coalition to Pave Way for First Cross-Atlantic H2 Shipment by 2026," Hydrogen Europe, https:// hydrogeneurope.eu/coalition-to-pave-way-for-first-cross-atlantic-clean-h2-shipment-by-2026/.

- 217 Bergmann et al., "Energy Security and Geopolitics Conference."
- 218 "International Trade in Products Related to Green Energy," Eurostat, October 2023, https://ec.europa. eu/eurostat/statistics-explained/index.php?title=International_trade_in_products_related_to_green_ energy&oldid=579764#Wind_turbines:_China_largest_import_partner.2C_United_Kingdom_largest_ export_partner.
- 219 Wind Europe, "EU Starts Investigation Into Chinese Wind Turbines under New Foreign Subsidies Regulation," Press release, April 9, 2024, https://windeurope.org/newsroom/press-releases/eu-startsinvestigation-into-chinese-wind-turbines-under-new-foreign-subsidies-regulation/; and Melissa Eddy, "Germany's Solar Panel Industry, Once a Leader, Is Getting Squeezed," *New York Times*, March 25, 2024, https://www.nytimes.com/2024/03/25/business/germany-solar-panels-china-protectionism.html.
- 220 "EU's Response to the US Inflation Reduction Act (IRA)," European Parliament, June 2, 2023, https://www.europarl.europa.eu/thinktank/en/document/IPOL_IDA(2023)740087.
- 221 Bergmann et al., "Energy Security and Geopolitics Conference."
- 222 "Critical Raw Materials Act," European Council, May 24, 2024, https://www.consilium.europa.eu/en/ infographics/critical-raw-materials/.
- 223 "China's Sunwoda Plans \$274.7 Million Hungarian Battery Plant," Reuters, July 27, 2023, https://www.reuters.com/technology/chinas-sunwoda-plans-2747-mln-hungarian-battery-plant-2023-07-27/.
- 224 Stephen Stapczynski and Ruth Liao, "Qatar Delays LNG Cargoes to South Europe amid Red Sea Crisis," Bloomberg, January 24, 2024, https://www.bloomberg.com/news/articles/2024-01-24/qatar-delays-lngshipments-to-europe-amid-red-sea-conflict?sref=B2BBHw9t.
- 225 Gonzalo Escribano, Lara Lázaro Touza, and Ignacio Urbasos Arbeloa, *Revamping the Euro-Mediterranean Energy and Climate Space* (Madrid: Real Instituto Elcano, July 2023), https://www.realinstitutoelcano. org/en/policy-paper/revamping-the-euro-mediterranean-energy-and-climate-space/; and Gabriel Gavin and Gregorio Sorgi, "Calls Grow for EU Sanctions on Gas-Rich Azerbaijan over Ethnic Cleansing Fears," *Politico*, September 20, 2023, https://www.politico.eu/article/eu-sanctions-gas-deal-azerbaijan-grow-ethnic-cleansing-fears-nagorno-karabakh/.
- 226 Bergmann et al., "Energy Security and Geopolitics Conference."
- 227 European Commission, "Commission Sets Out Actions to Accelerate the Roll-out of Electricity Grids," Press release, November 28, 2023, https://ec.europa.eu/commission/presscorner/detail/en/ip_23_6044; and European Commission, "Commission Proposes 166 Cross-Border Energy Projects for EU Support to Help Deliver the European Green Deal," Press release, November 28, 2023, https://ec.europa.eu/ commission/presscorner/detail/en/IP 23 6047.
- 228 "Connecting Europe Facility," European Commission, https://cinea.ec.europa.eu/programmes/ connecting-europe-facility_en.
- 229 Nadja Skopljak, "Construction Begins on First Subsea Interconnector between Spain and France," Offshore Energy, October 6, 2023, https://www.offshore-energy.biz/construction-begins-on-firstsubsea-interconnector-between-spain-and-france/; and Directorate-General for Energy, "Estonia, Latvia & Lithuania Agree to Synchronise Their Electricity Grids with the European Grid by Early 2025," European Commission, August 3, 2023, https://energy.ec.europa.eu/news/estonia-latvia-lithuania-agreesynchronise-their-electricity-grids-european-grid-early-2025-2023-08-03_en.
- 230 Gonzalo Escribano et al., "An Energy Union without Interconnections? Public Acceptance of Cross-Border Interconnectors in Four European Countries," *Energy* 266 (March 1, 2023): 126385, https://www. sciencedirect.com/science/article/pii/S0360544222032716.

- 231 Elisabeth Cremona and Chris Rosslowe, *Putting the Mission in Transmission: Grids for Europe's Energy Transition* (London: EMBER, March 2024), https://ember-climate.org/insights/research/putting-themission-in-transmission-grids-for-europes-energy-transition/.
- "About the Connecting Europe Facility," European Commission, https://cinea.ec.europa.eu/
 programmes/connecting-europe-facility/about-connecting-europe-facility_en#cef-energy; and
 "Commission Sets out Actions to Accelerate the Roll-out of Electricity Grids," European Commission,
 November 28, 2023, https://ec.europa.eu/commission/presscorner/detail/en/ip_23_6044.
- 233 European Climate, Infrastructure and Environment Executive Agency, "CEF Energy: The Biscay Gulf Electricity Interconnection Moves Forward," European Commission, April 17, 2023, https:// cinea.ec.europa.eu/news-events/news/cef-energy-biscay-gulf-electricity-interconnection-movesforward-2023-04-17_en.
- 234 "Electricity Interconnection Targets," European Commission, https://energy.ec.europa.eu/topics/ infrastructure/electricity-interconnection-targets_en.
- 235 Matia Riemer, Florian Schreiner, and Jakob Wachsmuth, *Conversion of LNG Terminals for Liquid Hydrogen or Ammonia* (Karlsruhe, Germany: Fraunhofer Institute for Systems and Innovation Research ISI, November 2022), https://www.isi.fraunhofer.de/content/dam/isi/dokumente/cce/2022/Report_ Conversion_of_LNG_Terminals_for_Liquid_Hydrogen_or_Ammonia.pdf.
- 236 "European Hydrogen Backbone Maps," European Hydrogen Backbone, https://ehb.eu/page/europeanhydrogen-backbone-maps.
- 237 Massimiliano Patierno, "H2Med Project: The New MidCat of Green Hydrogen," Instituto internacional de derecho y medio ambiente [International Institute for Law and the Environment], April 20, 2023, https://iidma.org/en/h2med-project-the-new-midcat-of-green-hydrogen/.
- 238 Andrius Sytas and Anne Kauranen, "Three Baltic Pipe and Cable Incidents 'Are Related,' Estonia Says," Reuters, October 27, 2023, https://www.reuters.com/world/europe/three-baltic-pipe-cable-incidents-arerelated-estonia-says-2023-10-27/.
- 239 Claudia Chiappa and Pierre Emmanuel Ngendakumana, "'Everything Indicates' Chinese Ship Damaged Baltic Pipeline on Purpose, Finland Says," *Politico*, December 1, 2023, https://www.politico.eu/article/ balticconnector-damage-likely-to-be-intentional-finnish-minister-says-china-estonia/.
- 240 Bergman et al., "Energy Security and Geopolitics Conference."
- 241 Sean Monaghan et al., *NATO's Role in Protecting Critical Undersea Infrastructure* (Washington, DC: CSIS, December 2023), https://www.csis.org/analysis/natos-role-protecting-critical-undersea-infrastructure.
- 242 "NATO Holds First Meeting of Critical Undersea Infrastructure Network," NATO, May 23, 2024, https://www.nato.int/cps/en/natohq/news 225582.htm.
- 243 Jeff Mason, Steve Holland, and Andrea Shalal, "United States, EU Agree to Start Talks on Critical Minerals amid Trade Tensions," Reuters, March 10, 2023, https://www.reuters.com/markets/commodities/amidtrade-dispute-us-eu-seek-minerals-agreement-talks-subsidies-2023-03-10; and "U.S.-E.U. Trade and Technology Council (TTC)," Office of United States Trade Representative, https://ustr.gov/useuttc.
- 244 "Citizen Support for Climate Action," European Commission, https://climate.ec.europa.eu/citizens/ citizen-support-climate-action_en.
- 245 "EU Challenges and Priorities in 2023," European Commission, September 2023, https://europa.eu/eurobarometer/surveys/detail/3092.

CSIS CENTER FOR STRATEGIC & INTERNATIONAL STUDIES

1616 Rhode Island Avenue NW Washington, DC 20036 202 887 0200 | www.csis.org